

Gender segregation: analysis across sectoral dominance in the UK labour market

Riccardo Leoncini^{1,2} · Mariele Macaluso¹ · Annalivia Polselli³

Received: 24 April 2023 / Accepted: 24 April 2024 / Published online: 21 June 2024 © The Author(s) 2024

Abstract

This paper aims to evaluate how changing patterns of sectoral gender segregation play a role in accounting for women's employment contracts and wages in the UK between 2005 and 2020. We then study wage differentials in gender-specific dominated sectors. We found that the propensity of women to be distributed differently across sectors is a major factor contributing to explaining the differences in wages and contract opportunities. Hence, the disproportion of women in female-dominated sectors implies contractual features and lower wages typical of that sector, on average, for all workers. This difference is primarily explained by "persistent discriminatory constraints", while human capital-related characteristics play a minor role. However, wage differentials would shrink if workers had the same potential and residual wages as men in maledominated sectors. Moreover, this does not happen at the top of the wage distribution, where wage differentials among women working in female-dominated sectors are always more pronounced than those among men.

Keywords Gender sectoral segregation \cdot Labour markets \cdot Gender inequality \cdot Wage differentials

JEL Classification $J16 \cdot J2 \cdot J31 \cdot J61 \cdot J71$

1 Introduction

Gender segregation across industrial sectors persists as a prevailing characteristic of the United Kingdom (UK)'s labour market (Government Equalities Office 2019; Irvine 2022), with the Covid-19 pandemic exacerbating this trend (Open Society Foundations

We are grateful to David Card for the helpful direction, especially related to the literature. We thank Emma Duchini, Marco Francesconi, Giovanni Guidetti and Giulio Pedrini for their helpful comments. We also thank conference participants at the AASLE 2022 Conference, 4th International Conference of European Studies, Summer School in Labor Economics at Barcellona School of Economics, 2023 Petralia Workshop. Our thanks go also to the editor and two anonymous reviewers for their detailed and useful comments. Annalivia Polselli acknowledges financial support from the South East Network for Social Sciences (SeNSS) as part of the Doctoral Training Partnership (award ES/P00072X/1/2128216).

Extended author information available on the last page of the article

2020; Johnston 2021). Between 2005 and 2020, the share of women in total employment in the UK exceeded 70% in sectors such as education, health and households as employers,¹ while it was below 30% in sectors like agriculture, mining and quarrying, manufacturing, construction and transport (Table 1). To support equal treatment of workers in the workplace and improve gender diversity across industries, the UK government has implemented several reforms in the past decade, including the Equality Act 2010 (EA2010), which sets out several measures prohibiting, among others, gender discrimination in employment and pay.² Despite these initiatives leading to a more balanced participation rate (Office for National Statistics 2022), the overrepresentation of women in certain industrial sectors is responsible for the sorting of women across occupations and disparities in both employment opportunities and wages (Olsen et al. 2018; Razzu and Singleton 2018).³

This paper uses the UK Labour Force Survey (LFS) quarterly data to investigate gender sectoral segregation in the UK between 2005 and 2020 by addressing two questions: (1) how gender sectoral segregation relates to the type of employment contracts (i.e. part-time, permanent, remote work, number of weekly working hours) and hourly wages; and (2) to what extent gender wage differentials differ in terms of observable and unobservable factors in female- and male-dominated sectors.

The first question is addressed through propensity score matching (PSM) by estimating the average differences in labour market outcomes between workers in femaleand male-dominated sectors with similar observed socio-demographic and working characteristics.

To answer the second question, we first use the threefold Kitagawa (1955)-Blinder (1973)-Oaxaca (1973) (KBO) decomposition to explore differences in wages due to human capital and productivity, or unexplained factors, or their simultaneous effect. We then estimate Mincer wage regression to explore the contribution of human capital and other observable skills. However, since there might be unobservable factors—e.g. behavioural traits such as self-esteem, ambition, competitiveness and willingness to make risky career choices (Gneezy et al. 2003; Gneezy and Rustichini 2004; Booth 2009; Bertrand 2011; Saccardo et al. 2018)—that may contribute to driving wage differentials, we retrieve predicted and residual wages from Mincer regression. We also conducted a counterfactual analysis to study how predicted and residual wages differ if workers (women in male- and female-dominated sectors and men in female-dominated sectors) had the same characteristics as men in male-dominated sectors.

Our main findings can be summarised in three points. First, gender-based sectoral segregation matters in the disparity of contractual opportunities, even controlling for

¹ The sector of *Households as Employers* includes the activities of domestic personnel (e.g. maids, cooks, waiters, gardeners, chauffeurs, caretakers, babysitters, etc.) whose product is consumed by the employing household (ONS: http://www.siccodesupport.co.uk/sic-division.php?division=97).

² According to the EA2010 and its related extensions (e.g. Regulations 2011 - Specific Duties and Public Authorities), a woman must not be discriminated against with respect to a man in a similar situation or due to a particular policy or working practice. For an updated map of the key policies regarding equality for and between women, see the UK Parliament website https://www.parliament.uk/globalassets/documents/ commons/scrutiny/gender_equality_policy_map_april_2020.pdf.

³ Tables A1 and A2 in the Appendix show that occupations in traditionally female-dominated sectors (e.g. education) are mainly over-represented by women, even in those occupations that are usually considered male-dominated (e.g. manager, director and senior officials, or skilled trades).

Table 1 Share of women, by sector a	nd year
-------------------------------------	---------

Sectors	Women's share (%)						
	2005	2010	2015	2020	2005-2020		
A—Agriculture, forestry and fishing	30.8	25.4	32.7	31.9	29.9		
B—Mining and quarrying	15.3	13.8	15.2	23.4	18.0		
C—Manufacturing	25.6	24.6	25.7	28.8	25.9		
D-Electricity, gas and air con supply	24.6	25.1	27.1	26.7	27.6		
E—Water supply, sewerage and waste	21.3	18.8	21.7	23.2	20.3		
F—Construction	14.6	16.8	18.4	21.1	16.8		
G—Distribution	53.4	51.5	51.2	49.3	51.7		
H—Transport and storage	26.4	23.0	25.1	24.8	24.5		
I—Accommodation and food services	58.6	57.9	56.0	57.9	57.5		
J—Information and communication	27.8	30.9	29.5	32.6	30.4		
K—Financial and insurance services	54.2	51.0	50.1	48.3	50.9		
L—Real estate services	56.2	63.2	55.3	58.3	57.8		
M—Professional, scientific and technical activities	49.8	47.7	48.2	46.8	48.0		
N—Admin and support services	24.5	46.9	49.7	48.8	44.6		
P—Education	74.3	75.6	74.8	76.1	75.3		
Q-Health and social work	80.5	80.6	80.3	79.1	80.4		
R—Arts, entertainment and recreation	50.0	52.4	51.3	50.7	50.3		
S—Other service activities	66.1	61.7	61.8	60.6	62.4		
T—Households as employers	68.6	78.4	79.6	77.1	74.8		

Sectors labelled as "O—Public admin and defence" and "U—Extra territorial" are removed from the sample because their contracts and wages highly differ from other sectors. The share is calculated over total employment, whereas gender sectoral dominance is calculated with the Sectoral Segregation Index in Eq. (2)

occupational composition. Workers in female-dominated sectors are more likely to be segregated into atypical contracts (part-time), to work fewer hours and less from home and to earn less than their counterparts in male-dominated sectors. The penalty for men working in those sectors is even larger than for women. Second, human capital and background characteristics play a minor role in explaining gender wage differentials. Instead, most of the difference is driven by persistent "discriminatory constraints" (Altonji and Blank 1999), such as barriers in the labour market for women due to the effects of discrimination (i.e. unequal pay for equally qualified workers) and unobserved differences in productivity and tastes. Third, wage differentials in female- and male-dominated sectors would shrink if workers had the same potential and residual wages as men in male-dominated sectors. However, women in femaledominated sectors would always earn less than men in high-paid jobs due to the negative selection in the labour market.

While most of the literature focuses on the role of gender segregation in explaining the gender wage gap by looking at occupational and job dimensions (Blackburn et al. 1993; Watts 1992, 1995, 1998; Petrongolo 2004; Cortés and Pan 2018; Folke and

Rickne 2022; Scarborough et al. 2021),⁴ our work is closely related to the scant literature on the role of gender segregation across sectors (Moir and Smith 1979; Kreimer 2004; Campos-Soria and Ropero-García 2016; Kamerāde and Richardson 2018; Scarborough et al. 2021). These papers highlight how gender division of labour is still embedded in sectors (Carvalho et al. 2019), which is considered a structural factor shaping the differential effects on labour markets caused by economic recessions (Rubery 2010; Rubery and Rafferty 2013; Kamerāde and Richardson 2018) and the business cycle (Hoynes et al. 2012; Périvier 2014; Doepke and Tertilt 2016; Razzu et al. 2020; Piłatowska and Witkowska 2022). For instance, Olivetti and Petrongolo (2014) emphasise the interplay between gender trends and the evolution of the industry structure. Therefore, an understanding of the sectoral composition of the workforce is necessary to assess the trajectory of male and female employment and wages (Moir and Smith 1979).⁵ Our contribution focuses on three main points. First, an innovative feature of this study is that it goes beyond the standard segregation index to identify female- and male-dominated sectors by further classifying sectors into high- and lowsegregated. This index allows us to measure the degree of imbalance and the intensity of differences in gender distribution of a sector over time. Second, our analysis is not limited to the KBO results that show which effect drives wage differentials. Instead, we disentangle the effect of differences in average characteristics of male and female employees and the effect of selection into different sectors of women and men on the gender wage differentials. Third, since unobserved confounders may enter the decision to choose a specific (female or male) sector, we address this potential selection bias by calculating the predicted and residual wages from the Mincer regression. This approach is similar to the method used in the literature on migration to calculate individual potential earnings (Parey et al. 2017) and capture the part of earnings that is uncorrelated to observed skills (Gould and Moav 2016; Borjas et al. 2019).⁶

Finally, we extend the findings of the 1980s literature on the issue of "comparable worth"⁷ (Treiman and Hartmann 1981; Maahs et al. 1985; Bielby and Baron 1986; Aaron and Lougy 1987). This literature found that the disproportion of women in

⁴ These studies found a threefold explanation for gender differences: (i) women's preferences for more flexible and family-oriented contracts (Petrongolo 2004; Bertrand 2011; Goldin 2014; Bertrand 2020; Morchio and Moser 2021) and less competitive and risky environments (Gneezy et al. 2003; Saccardo et al. 2018); (ii) comparative advantage in terms of human capital and productivity (Petrongolo 2004; Pető and Reizer 2021); (iii) discrimination (Petrongolo 2004) and sexual harassment (Folke and Rickne 2022).

⁵ The gendered division of labour across sectors, mainly due to persistent social norms and stereotypes, influences how women self-select into different jobs and careers and bargain their contracts (Card et al. 2016), and is likely to distort preferences, labour market trajectories and future wages (Mumford and Smith 2008; Reuben et al. 2017; Cortés and Pan 2020; Jewell et al. 2020; Card et al. 2021).

⁶ This literature highlights that immigrants could be positively/negatively selected based on both observed (e.g. higher levels of education) and unobserved determinants of labour market success (e.g. motivation, ambition and ability) that can enter into the decision to self-select into migration (Chiswick 1978, 1986, 1999; Borjas 1987; Bertoli et al. 2016).

⁷ Comparable worth, or "the women's issue of 1980", was a wage-setting policy on a firm-by-firm basis proposed to reduce the gender gap in earnings. Accordingly, jobs within a firm with comparable worth should receive equal compensation (Ehrenberg and Smith 1987). This was based on job evaluation scores to compare jobs in different occupations (Madden 1987). However, this implementation was not without disagreements (e.g. United States Commission on Civil Rights 1984; Gleason 1985; Ferber 1986; Aaron and Lougy 1987; Gerhart 1991).

female-dominated occupations is associated with lower pay in that occupation, on average, for all employees—men and women (Treiman and Hartmann 1981; Killingsworth 1987). However, the negative effect on the wage of being in such jobs is more significant for men than women (Roos 1981), even after controlling for relevant worker and job characteristics, including industry effects (Johnson and Solon 1984). We find that these negative results—i.e. worst job characteristics and wages for both men and women in female-dominated environments—are confirmed even when looking at industrial sectors. However, we found a more pronounced wage differential among women than men in female-dominated sectors at the top of the wage distribution.

The rest of the paper is structured as follows. Section 2 discusses the measures of gender sectoral dominance and segregation. Section 3 describes the data and reports some descriptive analysis. Section 4 presents the empirical strategy. Section 5 reports the estimated results. Section 6 concludes.

2 Conceptual framework

2.1 Gender sectoral segregation index

Gender sectoral segregation arises when a disproportionate share of men or women exists in a sector of the economy, independent of the nature of the job allocation (Watts 1998). In this section, we introduce the notion and measures of gender segregation in industrial sectors.

A sector is *female-dominated* (*fml-dom*) if the share of women employed in that sector is higher than the share of men in the same sector; it is *male-dominated* (*ml-dom*) if the share of men is higher than the share of women in the same sector. In formulae, the classification criterion for gender sectoral dominance is as follows:

Sectoral Dominance =
$$\begin{cases} \text{Female} & \text{if } \frac{W_{jt}}{W_t} > \frac{M_{jt}}{M_t} \\ \text{Male} & \text{otherwise} \end{cases}$$
(1)

where W_{jt} and M_{jt} are, respectively, the total number of women and men employed in sector *j* (SIC 1-digit) at time *t*; W_t and M_t are, respectively, the total number of female and male workers at time *t*.

The classification criterion defined in (1) uses the "majority voting" rule—i.e. the group with the largest number of members (either male or female) represents the sector.⁸

Based on classification criterion (1), we define the Sectoral Segregation Index (SSI_t^s) as a measure of the degree of disproportion in the distributions of men and women in female- and male-dominated sectors at each time period. The index is based on the well-

⁸ The denominators in (1) are not total employment (i.e. male plus female employees) but total employment by gender group, providing the overall share of women (or men) in a sector. The advantage of the criterion consists in avoiding the use of conservative thresholds of more than 60% (Killingsworth 1987) to classify a sector as female-dominated.

known Index of Dissimilarity,⁹ which is used in labour (Watts 1998) and education economics (Zoloth 1976; James and Taeuber 1985) to study group composition and quantify the segregation among two groups (Cortese et al. 1976).

 SSI_t^s is calculated for the two gender-dominated sectors ($SSI^{fml-dom}$ and SSI^{ml-dom}) as follows:

$$SSI_t^s = \frac{1}{2} \sum_{j \in J_s} \left| \frac{W_{jt}}{W_t} - \frac{M_{jt}}{M_t} \right| \text{ for all } t \text{ and } s = \{\text{ml-dom, fml-dom}\}$$
(2)

where J_s is the set of sectors in a male-dominated or female-dominated group. The index informs on time-varying group imbalance within gender-dominated sectors and ranges between 0 and 1. Large values of SSI^{fml-dom} (or SSI^{ml-dom}) flag large gender imbalance towards women (or men) and indicate the proportion of women (or men) that would have to either leave or enter each sector to avoid gender sectoral segregation. The value of the index remains unchanged when transferring workers between sectors within each gender group.¹⁰

Because the difference in the share of female and male employees can be extremely low in some sectors or high in others, we define an additional index that allows us to distinguish between high- and low-segregated sectors. Specifically, on average, sectors that display low (or high) segregation are classified as low (or highly) segregated sectors by ranking them from the least to the most segregated, based on SSI³_t.

Our analysis follows the UK Standard Industrial Classification (SIC) at one digit, which is used for business establishments by type of economic activity. We consider the 19 sectors listed in Table 2 as female- or male-dominated sectors and by the degree of segregation (high and low).¹¹

Figure 1 shows the evolution of SSI_t^s over 2005–2020 in the UK economy (upper graph), in low-segregated (bottom left graph) and high-segregated sectors (bottom right graph). The index follows a downward trajectory in male-dominated sectors (from 0.182 in 2005 to 0.158 in 2020), meaning that sectoral segregation has decreased. Instead, the segregation remains stable at around 0.17 over time in female-dominated sectors. In low-segregated sectors, gender segregation is larger in female-dominated than male-dominated sectors (with mean values of 0.021 and 0.011, respectively). Instead, in high-segregated sectors, gender segregation is higher in male-dominated sectors. However, it decreased over time (from 0.35 in 2005 to 0.30 in 2020) and dropped below the level of segregation in female-dominated sectors after 2015.

⁹ The Index of Dissimilarity measures the degree of the disproportion in the distributions of two groups. It provides information on the proportion of the minority group that would have to be transferred to reach no segregation (Cortese et al. 1976; Zoloth 1976; Watts 1998). Duncan and Duncan (1955)'s seminal article provides the first systematic review of segregation indices. A more comprehensive approach to segregation measurement has been available since the 1980's (Reardon and Firebaugh 2002).

 $^{^{10}}$ A similar interpretation is provided by Zoloth (1976) to describe the racial composition of schools within and across districts. In her setting, the Dissimilarity Index (*D*) remains unaffected by transferring students between schools within each group (minority and non-minority students). *D* changes only by transferring students across the two groups.

¹¹ For years before 2008, we used the correspondence between SIC 2003 and SIC 2007 sections. Sectors labelled as O—Public administration and defence and U—Extra territorial are removed from the sample due to the different nature of contracts and wages in their related jobs.

Table 2 List of high- and low-segregated sectors

High segregated sectors	Low segregated sectors
Female-dominated sectors	
I—Accommodation and food services	G—Distribution
P—Education	L—Real estate services
Q—Health and social work	S—Other service activities
	T—Households as employers
Male-dominated sectors	
B—Mining and quarrying	A—Agriculture, forestry and fishing
C—Manufacturing	K-Financial and insurance services
D—Electricity, gas and air con supply	R-Arts, entertainment and recreation
E—Water supply, sewerage and waste	
F—Construction	
H—Transport and Storage	
J—Information and communication	
M—Professional, scientific and technical activities	
N—Admin and support services	

Sectors labelled as "O—Public admin and defence" and "U—Extra territorial" are removed from the sample because their contracts and wages highly differ from other sectors

3 Data and descriptive statistics

3.1 Data source and characteristics of the sample

Our analysis is based on the Labour Force Survey (LFS) quarterly data from the UK Office for National Statistics (ONS). LFS is the most extensive household study in the UK, providing a comprehensive source of data on workers and the labour market. The analysis spans from fiscal years between 2005 and 2020. This period covers widespread enforcement of equality legislation, the 2007–2008 financial and economic crisis, and the recent changes caused by the COVID-19 outbreak.¹²

We focus on working-age (16–64) employees, i.e. people who are in employment and paid a wage by an employer for their work.¹³ Our final sample consists of 334,055 female workers and 307,245 male workers. The dataset includes variables on a wide range of (i) demographic characteristics (gender, age, country of birth, nationality, ethnicity, religion); (ii) socio-economic factors (presence of dependent children, marital status, education, experience, full/part-time job, remote work, public sector, training

¹² Most of the literature shows that the 2007–2008 crisis had a severe impact on male-dominated sectors, such as on construction and manufacturing (Hoynes et al. 2012; Périvier 2014; Doepke and Tertilt 2016). In contrast, the COVID-19 crisis has hit counter-cyclical sectors (e.g. in-person services) sharply (Piłatowska and Witkowska 2022).

¹³ We excluded self-employed workers (i.e. people who, in their main employment, work on their account whether or not they have employees) from the analysis because the earnings questions are not addressed to respondents who are self-employed in the UK LFS. This practice is standard in this type of analysis (see, e.g. Dustmann et al. 2010; Dustmann and Frattini 2014).





Fig. 1 Evolution of SSI index over fiscal year

opportunities, sectors and occupations); (iii) geographical information on residence and working region. Information on wages in the LFS is the self-reported gross weekly pay for the reference week.¹⁴ The number of hours is the total usual hours worked in the main job per week, including usual hours of paid overtime when applicable.

Table 3 reports the summary statistics of the main variables by gender. Regarding demographic characteristics, there is a substantial prevalence of UK natives in both male and female samples (87%), followed by non-European Economic Area (non-EEA) immigrants and EEA citizens. The average age is similar for men and women (around 40 years). On average, women in the sample are as educated as men (13 years of education) and have slightly less experience (22.99 years of experience against 23.46). More than half of the women and men are either married or cohabiting. In addition, 46% of women and 42% of men have dependent children under 18 years.

Segregation variables show that 69% of women are employed in female-dominated sectors compared to 35% of men. Around 26% of both women and men work in low-segregated sectors, while the rest work in highly segregated sectors.

Focusing on the outcomes of interest, both female and male employees have the same share of permanent jobs (around 94%). Women's wages (in logarithm) are, on average, lower than men's (2.41 percentage points against 2.59 for men). A small share of employees work from home (5% of the women against 8% of the men). On average, women work 31 hours per week and 42% part-time, while men work 40 hours per week and 10% part-time. A more detailed investigation of the reasons for working part-time in Table 4 highlights that the share of women who *did not want* a full-time job is much larger than those who *could not find* it (78% against 10%). In contrast, 39% of men *did not want* a full-time job and 26% *could not find* one. Family duties and domestic commitments are among the main reasons for not wanting a full-time job (48% and 28%, respectively). Instead, men do not want a full-time job mainly because *they are financially secure and work because they want* (24%) or for other reasons (34%).

3.2 Shift-share decomposition of employment

This section provides a descriptive picture of gender trends in employment in maleand-female-dominated sectors. We adopt a revised version of Olivetti and Petrongolo's (2016) shift-share decomposition.¹⁵

In our analysis, the growth of female employment share is decomposed into (i) the change in the total *employment share* of the sector (*between component*) and (ii) the

¹⁴ We calculate the real wage based on hourly wages in 2015 prices as real wage = hour pay/(CPI2015/100).

¹⁵ Unlike the original paper that uses the number of worked hours, we use the employment shares. Razzu et al. (2020) present an extension of Olivetti and Petrongolo's (2016) decomposition considering the role of changing types of employment within industry sectors according to education from 1971 to 2016 in the UK.

Variable	Women		Men		Full sample		
	Mean	Std. dev.	Mean	Std. dev.	Mean	Std. dev.	
Demographic characteristics							
Natives	0.87	0.33	0.87	0.34	0.87	0.33	
EEA	0.05	0.22	0.05	0.22	0.05	0.22	
Non-EEA	0.07	0.26	0.08	0.27	0.08	0.27	
Age	40.58	11.99	41.15	12.37	40.85	12.18	
Black	0.02	0.14	0.02	0.13	0.02	0.14	
Asian	0.03	0.17	0.04	0.20	0.04	0.18	
Other ethnicity	0.03	0.16	0.03	0.16	0.03	0.16	
Muslim	0.01	0.12	0.03	0.16	0.02	0.14	
Christian	0.61	0.49	0.54	0.50	0.57	0.50	
Other religions	0.14	0.35	0.14	0.34	0.14	0.35	
Socio-economic factors							
In couple	0.55	0.50	0.58	0.49	0.57	0.50	
With dependent children (<18yrs)	0.46	0.50	0.42	0.49	0.44	0.50	
Years of education	13.77	3.07	13.58	2.91	13.68	2.99	
Experience	22.99	12.44	23.46	12.83	23.21	12.63	
Training	0.34	0.47	0.33	0.47	0.34	0.47	
Log wages	2.41	0.50	2.59	0.56	2.50	0.54	
Part-time work	0.42	0.49	0.10	0.29	0.26	0.44	
Public sector	0.34	0.47	0.14	0.35	0.24	0.43	
Permanent job	0.94	0.23	0.95	0.21	0.95	0.22	
Weekly hours	30.77	12.87	40.19	12.53	35.28	13.55	
Log weekly hours	3.32	0.51	3.64	0.37	3.47	0.48	
Remote work	0.05	0.21	0.08	0.26	0.06	0.24	
Benefit	0.40	0.49	0.11	0.31	0.26	0.44	
Female dominance	0.69	0.46	0.35	0.48	0.53	0.50	
Low segregation	0.27	0.44	0.26	0.44	0.26	0.44	

Table 3 Summary statistics for employed workers

The descriptive statistics in the table are for the sample of employed workers (self-employed are excluded from the sample). Total number of female workers is 334,055. Total number of male workers is 307,245. Total number of workers in the sample is 641,300

change in gender composition within the sector (within component):

$$\Delta e_{st}^{g} = \underbrace{\sum_{j=1}^{J_{s}} \alpha_{jt}^{g} \Delta e_{jt}}_{\text{Between-sector}} + \underbrace{\sum_{j=1}^{J_{s}} \alpha_{jt} \Delta e_{jt}^{g}}_{\text{Within-sector}} \quad \text{for } s = \{\text{fml-dom, ml-dom}\}, g = \{\text{fml, ml}\}_{(3)}$$

Deringer

	Women (%)	Men (%)	Full sample (%)
Reasons for part-time work			
Student or at school	9.69	30.57	13.33
Ill or disabled	2.15	4.65	2.59
Could not find full-time job	9.95	25.88	12.74
Did not want full-time job	78.21	38.91	71.34
Total	100	100	100
Reasons for not wanting full-time job			
Financially secure—work because want	7.22	24.26	8.74
Earn enough part-time	7.64	17.41	8.52
Want to spend more time with family	40.35	12.30	37.85
Domestic commitments prevent full-time	28.20	10.70	26.64
Insufficient child-care facilities	3.49	0.96	3.26
Another reason	13.09	34.37	14.99
Total	100	100	100

 Table 4
 Reasons for part-time work

Percentages (%) are over group total

where $\Delta e_{st}^g = \frac{E_{st}^g}{E_{st}} - \frac{E_{l_0}^g}{E_{l_0}}$ is the difference in the share of female/male employment between the base period t_0 and the current period t; $\Delta e_{jt} = \frac{E_{jt}}{E_t} - \frac{E_{jt_0}}{E_{l_0}}$ is the difference in the share of total employment in sector j between t_0 and t; $\Delta e_{jt}^g = \frac{E_{jt}^g}{E_{jt}} - \frac{E_{jt_0}^g}{E_{jt_0}}$ is the difference in the share of female/male employment in sector j; $\alpha_{jt}^g = \frac{\left(e_{jt_0}^g + e_{jt}^g\right)}{2}$ and $\alpha_{jt} = \frac{\left(e_{jt_0} + e_{jt}\right)}{2}$ are decomposition weights (i.e. the average share of female employment in sector j and the average share of sector j, respectively). The reference year is the first available year in the dataset ($t_0 = 2005$); s stands for sectors classified as female-/male-dominated according to Eq. (1).

Figure 2 displays the shift-share decomposition of female and male employment (left and right graphs, respectively). The graphs show the difference in employment with respect to the base year (2005) and its decomposition into *between* and *within* components (respectively, dashed and dotted lines) in female- and male-dominated sectors (crosses and circles).

The 2007–2008 financial and economic crisis harshly hit male employment. In particular, male employment share (*between* component) and male composition (*within* component) in male-dominated sectors suddenly decreased. Conversely, the crisis stimulated female employment, not only in the total employment share (both *between* components of female- and male-dominated sectors) but also in the composition of women in female-dominated sectors (*within* component). The COVID-19 outbreak arrested the overall female employment in both male and female-dominated sectors (*between* components). As expected, this led to a reduction in the composition of



Fig. 2 Shift-share decomposition, by gender. *Note:* The graph shows the difference in employment in the comparison year with respect to the base year (i.e. the fiscal year 2005). The overall change in employment is shown in solid line and its decomposition into the 'between' and 'within' components, respectively, with dashed and dotted lines. The cross marks the components for female-dominated sectors and the circle the components for male sectors. The 'between' component (BTW) captures the change due to changes in the sectoral structure of the economy; the 'within' component (WTHN) reflects changes in female composition within sectors

women in female-dominated sectors (*within* component) but to a sharp increase in the composition of women in male-dominated sectors.¹⁶

Overall, female composition (*within* component) increased gradually in maledominated sectors after 2010. It seems that EA2010 stimulated female employment from the demand side. This may suggest that a higher proportion of women were employed *within* each male-dominated sector at the expense of decreasing male employment.

4 Empirical strategy

4.1 Estimating gender sectoral segregation on employment contracts and wages

In this section, we outline the methodology employed to compare the average difference in labour market outcomes (permanent jobs, part-time jobs, working hours, remote work and hourly wages) in male- and female-dominated sectors among workers with similar observable skills and socio-demographic characteristics. We employ the propensity score matching (PSM) method for this purpose. This approach constructs a counterfactual group by matching workers in female-dominated sectors with those working in male-dominated sectors based on the same *propensity score* (i.e. the estimated probability of being employed in a female-dominated sector conditional on the observed characteristics). The estimand of interest is the Average Treatment

¹⁶ These results are in line with the literature assessing that during the recession period of the 2007–2008 crisis, female employment was generally affected less than male employees, while during the recovery phase, male employment recovered faster than female employment (Hoynes et al. 2012; Doepke and Tertilt 2016; Ellieroth et al. 2019; Albanesi and Kim 2021). For instance, Ellieroth et al. (2019) finds that married women are more stuck in employment during recessions. Therefore, their labour supply decisions account for the higher risk of job loss experienced by their husband. However, Razzu et al. (2020) emphasises how gender segregation across industrial sectors and occupations in the UK exacerbates women's employment and pay gap during the business cycle.

effect on the Treated (ATT) which is computed as the average of the difference between the *observed* outcomes (Y_1) and the *imputed* outcomes (Y_0) for each worker in female-dominated sectors, $\mathbb{E}\{Y_{1i} - Y_{0i} | \text{Working in fml-dom sector}\}$. The imputed outcomes for each worker in female-dominated sectors are calculated by using the average observed outcomes of similar individuals working in male-dominated sectors. The underlying assumption is that those who choose to work in female- and male-dominated sectors only differ in the endowment of their observed skills and human capital accumulation.

In Sect. 5.1, we compare the average conditional outcomes *between* gender sectoral dominance for the full sample (male and female workers together), male sample and female sample. We estimate the propensity scores via a Probit regression. For the selection of the covariates to calculate the propensity scores, we consider all factors associated with *working in female-dominated sector* (including interactions and squares), and then we use an automatic selection procedure—i.e. Least Absolute Shrinkage and Selection Operator (LASSO). Table A3 in the Appendix reports the selected covariates from the penalised regressions. Then, we conduct the sensitivity analysis to test whether the balancing property of the covariates before and after the match holds. The covariates are balanced if the standardised bias after matching is within $\pm 5\%$ (Rosenbaum and Rubin 1985). The matching method successfully builds a meaningful *comparison* group if the condition is satisfied.

4.2 Estimating wages in gender-specific dominated sectors

We now focus on the gendered differences in hourly wages in male- and femaledominated sectors based on observable and unobservable characteristics. For this purpose, we first perform the counterfactual Kitagawa–Blinder–Oaxaca (KBO) decomposition (Kitagawa 1955; Blinder 1973; Oaxaca 1973), which is an established method in the literature on discrimination (e.g. Mueller and Plug 2006; Blau and Kahn 2017) to study the difference in wages between women and men. We then run Mincer wage regression to explore the role of the average characteristics of male and female employees in female- and male-dominated sectors and to obtain the predicted and residual wages.

4.2.1 Decomposing the gender wage differentials

In this analysis, we use the threefold version of the KBO decomposition (Jann 2008), which decomposes the average difference of hourly wages (in logarithm) between men and women working in female- and male-dominated sectors into three components as follows¹⁷:

¹⁷ We also implemented a standard twofold KBO decomposition, which decomposes the wage differentials into a part that is explained by differences in human capital and productivity and an unexplained part usually interpreted as discrimination (Blau and Kahn 2017). We obtained the same results for the explained and the unexplained components. However, because the unexplained component is the algebraic sum of the coefficient and the interaction effects of the threefold decomposition, the twofold is less informative (see also Meara et al. 2020).

$$\underbrace{\mathbb{E}(y_{ml}^{s}) - \mathbb{E}(y_{fml}^{s})}_{\text{overall difference}} = \underbrace{\left[\mathbb{E}(\mathbf{X}_{ml}^{s}) - \mathbb{E}(\mathbf{X}_{fml}^{s})\right]' \boldsymbol{\beta}_{fml}^{s}}_{\text{endowment effect}} + \underbrace{\mathbb{E}\left(\mathbf{X}_{fml}^{s}\right)' (\boldsymbol{\beta}_{ml}^{s} - \boldsymbol{\beta}_{fml}^{s})}_{\text{coefficients effect}} + \underbrace{\left[\mathbb{E}(\mathbf{X}_{ml}^{s}) - \mathbb{E}(\mathbf{X}_{fml}^{s})\right]' \left(\boldsymbol{\beta}_{ml}^{s} - \boldsymbol{\beta}_{fml}^{s}\right)}_{\text{interaction effect}}$$
(4)

where **X** is a vector containing the covariates, such as socio-demographic variables, human capital variables and work-related variables; and β is a vector of slope parameters and the intercept; fml stands for women and ml for men, and $s = \{\text{fml-dom}, \text{ml-dom}\}.$

The *first component* explains observable group differences in the predictors, such as background and human capital characteristics of workers (*endowment effect*). This effect quantifies the expected change in women's wages if they had men's characteristics. A negative *endowment effect* shows that female workers possess better predictors than their male counterparts.

The *second term* explains differences in the coefficients, including the intercept, that arise from discrimination—i.e. unequal pay for equally qualified workers (Blau and Kahn 2017)—or cannot be explained by differences in the observed factors (*coefficient effect*). Specifically, the *coefficient effect* measures the expected change in the average wage of women if they had the coefficients of men. The intercept included in the effect captures the contribution of unobservable characteristics (Cotton 1988)—e.g. behavioural traits, such as self-esteem, ambition, competitiveness and the willingness to take risky career choices (Gneezy et al. 2003; Gneezy and Rustichini 2004; Bertrand 2011; Saccardo et al. 2018). A negative intercept term is interpreted as "ongoing discriminatory constraints", such as barriers in the labour market for the minority group due to the effects of discrimination and unobserved differences in productivity and tastes (Altonji and Blank 1999). When the overall *coefficient effect* is positive, women would have higher average wages if paid like men.

The *third component* explains the coexistence of differences in the endowments and coefficients between the two groups (*interaction effect*). If the *interaction effect* is positive, women have a "double disadvantage" because they have smaller coefficients than men when they have worse predictors; if it is negative, differences in coefficients and covariate levels offset each other (Jann 2018).

After assessing which effect drives the wage differences, in the following paragraphs, we investigate the contribution of each observable factor and unobservable characteristics that contribute to explaining the differences in wages between women and men in female- and male-dominated sectors.

4.2.2 The role of human capital

We use the Mincer regression to analyse the association of workers' human capital and observable skills with wage differentials between sectors and genders. The estimating

equation is as follows:

$$\mathbf{y} = \mathbf{X}\boldsymbol{\beta} + \delta_t + \boldsymbol{\epsilon} \tag{5}$$

where **y** is hourly wages in logarithm; **X** is $N \times k$ matrix of control variables (i.e. socio-demographic, human capital and work-related variables); and δ_t are the time fixed effects. Equation (5) is estimated using OLS.

The set of controls includes three groups of variables as follows. *Socio-demographic variables* include age and its square, nationality, ethnicity, religion, being in a stable relationship, having dependent children and the interaction of the last two. *Human capital variables* are education (low, intermediate and higher education), experience and its square, years in education and its square and training offered by the current employer.¹⁸ *Work-related variables* include a dummy for female-dominated sectors, a dummy for low gender sector segregation, a dummy for working in the public sector and the type of occupation. Working region dummies are included.¹⁹

Mincer regression is based on observable characteristics so as to "hold constant" individual factors that affect wages. However, this specification may not capture the selection of workers into sectors based on some relevant unobservable characteristics (e.g. self-esteem, ambition, competitiveness, risk aversion) that may influence wage differentials. In the next section, we address the possible bias that arises from omitting these unobservable factors.

4.2.3 Predicted and residual wages

We use the estimates of Mincer regression from Sect. 4.2.2 to measure how the selection on observable and unobservable characteristics shapes the difference in wages for men and women in female- and male-dominated sectors. First, we calculate the estimated returns to construct predicted wages, which measure the individual wage potential based on observable factors (Parey et al. 2017). Second, we follow Borjas et al. (2019) to shed light on the role of unobservable characteristics in the selection process by calculating residual wages, which capture the part of wages uncorrelated with workers' skills.

We consider four sub-groups from our sample of workers: men in male-dominated sectors (ml, ml-dom), women in male-dominated sectors (fml, ml-dom), men in female-dominated sectors (ml, fml-dom) and women in female-dominated sectors (fml, fml-dom). In addition, we conduct a counterfactual exercise to examine the trajectory of wage potentials and residuals for each sub-group if workers had the same estimated coefficients of men working in male-dominated sectors. In formulae,

$$\hat{\mathbf{y}}_{g,\text{gdom}}^{c} = \mathbf{X}_{g,\text{gdom}} \hat{\boldsymbol{\beta}}_{g,\text{gdom}}$$
(6)

$$\hat{\mathbf{u}}_{g,\text{gdom}}^{c} = \mathbf{y}_{g,\text{gdom}} - \hat{\mathbf{y}}_{g,\text{gdom}}^{c} \tag{7}$$

¹⁸ We included both the categorical variable for the education band (low, intermediate and high) and the continuous variable for years of education. The OLS assumption of the absence of perfect multicollinearity is not violated because years of education capture the intensity of the returns of education within each band.

¹⁹ Usual worked hours are not included in the specification because of possible endogeneity issues due to reverse causality. The estimates would be downward biased because hours would appear on both sides of the equation as wages are calculated based on usual worked hours per week (Borjas 1980).

where $g = \{ml, fml\}$ is the gender of the worker, and $gdom = \{ml-dom, fml-dom\}$ is the sector with a large share of the specified gender. This analysis allows us to compare how predicted and residual wages would differ if workers (women in female- and maledominated sectors and men in female-dominated sectors) had the same characteristics as the most advantaged group, i.e. men in male-dominated sectors.

Predicted and residual wages are sorted and used to construct the cumulative distribution functions (CDFs) by gender in female- and male-dominated sectors. Then, we compare the CDFs of men and women *between* and *within* gender sectoral dominance. The equality of the distributions of the (actual and counterfactual) predicted and residual wages among the four sub-groups is tested by using the nonparametric Kolmogorov–Smirnov (K–S) test.

5 Estimation results

5.1 Estimation results for the PSM on contracts and wages

In this section, we present the main results of the PSM by looking at three different samples (i.e. all workers, men and women). Working in a female-dominated sector is the treatment variable. Table 5 reports the ATT for each labour market outcome of interest—i.e. having a temporary job, part-time work, number of hours (in logarithm), remote work and wage (in logarithm).²⁰

The first result of the analysis is that contractual features usually associated with female workers are more common in female-dominated sectors, even among men. That is, workers in female-dominated sectors compared to their peers in male-dominated sectors have, on average, fewer permanent positions (respectively, 0.947 against 0.956), work more part-time (0.35 against 0.267), fewer hours (3.404 against 3.475) and less from home (0.037 against 0.90). This result remains valid even when looking at male and female workers separately. Specifically, if men and women in a female-dominated sector were hired in a male-dominated sector, they would have more permanent positions (0.9 p.p. and 0.7 p.p., respectively), would work more hours (7.7 and 6.3 p.p.), less part-time (8.4 and 7.5 p.p.) and more from home (4.3 and 5.9 p.p.). All ATT estimates are significantly different from zero at a 1% significance level.

The second main result is that there is a higher penalty for men than women working in female-dominated sectors, given their larger magnitudes of ATT (all significant at 1% level). This result is also confirmed when looking at wage differentials between female- and male-dominated sectors. Men in female-dominated sectors earn, on average, 15.4 p.p. less than their male peers in male-dominated sectors. Instead, women in female-dominated sectors earn, on average, 12.6 p.p. less than their female counterparts in male-dominated sectors. Overall, any worker in female-dominated sectors

²⁰ The propensity scores for matching workers employed in female-dominated sectors (fml-dom) with similar workers in male-dominated sectors (ml-dom) come from estimates reported in Table A4 in the Appendix. The table shows the likelihood of a worker being employed in a female-dominated sector based on socio-demographic characteristics and working environment. Being a woman in a stable relationship without dependent children decreases the probability of working in a female-dominated sector. Having dependent children, being non-European and working in operative jobs, technical and secretarial occupations reduces the likelihood of being in female-dominated sectors.

	2305

Variable	Working in:		Difference	S.E	T-stat	Units on support:		
	fml-dom	ml-dom	(ATT)			ml-dom	fml-dom	
Sample: Full								
Permanent	0.947	0.956	-0.009	0.001	- 10.44	295,963	321,014	
Part-time work	0.350	0.267	0.082	0.002	51.98	295,994	321,056	
ln(hours)	3.404	3.475	-0.071	0.002	- 43.64	296,025	321,094	
Remote work	0.037	0.090	- 0.053	0.001	-46.70	296,025	321,094	
ln(wage)	2.419	2.555	- 0.136	0.002	- 59.81	296,025	321,094	
Sample: Men								
Permanent	0.949	0.958	-0.009	0.001	- 9.51	195,607	100,778	
Part-time work	0.144	0.060	0.084	0.001	63.27	195,627	100,791	
ln(hours)	3.597	3.674	-0.077	0.002	- 50.52	195,650	100,802	
Remote work	0.049	0.092	- 0.043	0.001	- 33.90	195,650	100,802	
ln(wage)	2.502	2.656	-0.154	0.003	-58.40	195,650	100,802	
Sample: Women								
Permanent	0.947	0.954	-0.007	0.001	- 6.44	100,356	220,214	
Part-time work	0.444	0.369	0.075	0.003	28.27	100,367	220,243	
ln(hours)	3.315	3.378	- 0.063	0.003	- 24.51	100,375	220,270	
Remote work	0.031	0.090	- 0.059	0.002	- 39.18	100,375	220,270	
ln(wage)	2.380	2.506	- 0.126	0.003	- 42.40	100,375	220,270	

 Table 5
 Propensity score matching

S.E. does not take into account that the propensity score is estimated. The matching method is single nearest-neighbour; five neighbours are used to calculate the matched outcome. The matching algorithm imposes common support. The propensity scores obtained from Probit regression in Table A4 are used to match workers employed in female-dominated sectors (fml-dom) with similar workers in male-dominatesd sectors (ml-dom). The estimates are significantly different from zero at a 1% significance level

would be paid 13.6 p.p. more if employed in male-dominated sectors. These results are consistent with the findings of "comparable worth" literature, that is, jobs dominated by women pay, on average, less all employees (Treiman and Hartmann 1981; Killingsworth 1987), and the effect on wages in such jobs is more negative for men than women (Roos 1981; Johnson and Solon 1984).

The sensitivity analysis (Fig. B1 in the Appendix) confirms that the balancing property is satisfied for all samples since all covariates are well balanced. Therefore, the matching method effectively built a valid control group.

Overall, our analysis suggests that gender sectoral segregation is a relevant factor in explaining observed differences in employment contracts (i.e. part-time, permanent, remote work, number of weekly working hours) and wage differentials.



Fig.3 KBO decomposition, by gender sectoral dominance. ESTIMATION NOTE: Both models for women and men are estimated using the Mincerian regression equation (with OLS). The degree of gender segregation is not included because it is highly correlated with the grouping variable of gender sectoral dominance. The shaded areas are the 95% confidence intervals

5.2 Estimation results for wages

5.2.1 Results for the KBO

The evolution of the three components of the KBO decomposition and their sum over time is shown in Fig. 3. Women are contrasted to men within the same genderdominated sector. The dashed line represents the *coefficient effect*, the long-dashed line the *endowment effect* and the dotted line the part of the *interaction* component. The solid line is the sum of the three effects and reveals their overall contribution (for the contribution of each characteristics, see Tables A5, A6 in the Appendix).

The first result from the decomposition is that the difference in wages between men and women in both types of sectors is not so much explained by differences in human capital and productivity (*endowment effect*). The dynamics of the *endowment effect* shows that the gap in terms of observable characteristics has narrowed over time (the effect is close to zero), reflecting women's increased human capital levels relative to men's (as also observed by, for example, Goldin 2014; Blau and Kahn 2017). While men and women employed in female-dominated sectors are, on average, more similar in terms of human capital over time, the *endowment effect* is positive between 2010 and 2018 in male-dominated sectors, meaning that women have worse observed characteristics than men in those years. Instead, the most relevant result is that the difference is mostly due to "ongoing discriminatory constraints" in the labour market towards women (*coefficient effect*) stemming from substantial unexplained constraints in labour market returns. (The intercept is negative from Tables A5, A6 in the Appendix.) The *coefficient effect* is positive in both gender-dominated sectors, suggesting that women should be paid more than men to prevent any sort of discrimination for reasons other than human capital characteristics and productivity.²¹

The *interaction effect* explains little of the gender wage differential in both femaleand male-dominated sectors, although we observe a "double disadvantage" for women (positive *interaction effect*) in male-dominated sectors only before 2010.

5.2.2 Results based on human capital factors

As the KBO showed that human capital characteristics play a minor role in explaining wage differentials, the analysis in this section shows the contribution of each observable factor on wages. Table 6 reports the estimated coefficients of the Mincer wage regression by gender for all sectors (Columns 1–2) and gender-dominated sectors (Columns 3–6).

The estimates of the Mincer regression for segregation variables confirm the main result of the PSM. Specifically, working in female-dominated sectors is significantly negatively correlated with hourly wages for both men and women (-0.163 and -0.158, respectively). In addition, working in sectors with low gender sectoral segregation is significantly positively associated with higher wages for male workers in the full sample (0.027) but negatively correlated with wages for women in both female-and male-dominated sectors (-0.012 and -0.102, respectively). The interaction term between female-dominated sectors and low gender segregation is positive and significant for women only.

Focusing on human capital characteristics, workers with higher educational attainment earn, as expected, more than those with low education. More years of education are positively associated with wages but with a diminishing effect (the square is negative). Our calculations show that the optimal number of years in education that maximises wages is approximately 15.7 years for men as opposed to 19.5 years for women in the full sample.²² Therefore, women are expected to stay in education for more years than men, who need only a degree to earn their optimal wage. We obtained a similar number of years of education in female-dominated sectors (18 for women and 15.9 years for men), while the difference is less pronounced in male-dominated sectors (16.5 for women and 15.6 for men). Potential working experience has significant diminishing returns, and receiving training is significantly associated with an increase in the hourly wage, especially in male-dominated sectors.

²¹ However, it is worth mentioning that the *coefficient effect* may over-estimate the extent of discrimination if variables associated with human capital and individual preferences, which are negatively correlated with wages, are excluded from the regression equation (Cotton 1988; Altonji and Blank 1999; Blau and Kahn 2017).

²² The figures come from the following calculations: $0.157/(2 \times 0.005) = 15.7$ for men, and $0.117/(2 \times 0.003) = 19.5$ for women.

	Dep. var.: Log(Wage)						
	All sectors		Male-domin	nated sectors	Female-dor	ninated sectors	
	Man	Women	Man	Women	Man	Women	
	(1)	(2)	(3)	(4)	(5)	(6)	
Human capital variables							
Intermediate education	0.022***	0.020***	0.022***	0.025***	0.019***	0.017***	
	(0.003)	(0.002)	(0.004)	(0.005)	(0.004)	(0.003)	
High education	0.099***	0.100***	0.100***	0.103***	0.090***	0.099***	
	(0.006)	(0.005)	(0.007)	(0.010)	(0.009)	(0.006)	
Years of education	0.157***	0.117***	0.156***	0.132***	0.127***	0.108***	
	(0.003)	(0.003)	(0.004)	(0.006)	(0.006)	(0.003)	
Years of education ²	- 0.005***	- 0.003***	- 0.005***	- 0.004***	- 0.004***	- 0.003***	
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	
Experience	0.015***	0.013***	0.017***	0.016***	0.013***	0.012***	
	(0.001)	(0.001)	(0.001)	(0.002)	(0.001)	(0.001)	
Experience ²	- 0.000***	- 0.000***	- 0.000***	- 0.000***	- 0.000***	- 0.000***	
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	
Training	0.067***	0.049***	0.068***	0.063***	0.047***	0.032***	
	(0.002)	(0.002)	(0.003)	(0.003)	(0.003)	(0.002)	
Socio-demographic variable	?S						
Age	0.008***	0.004**	0.008***	0.011***	0.005*	- 0.003*	
	(0.001)	(0.001)	(0.002)	(0.003)	(0.002)	(0.001)	
Age ²	0.000***	0.000***	0.000*	0.000	0.000***	0.000***	
-	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	
EEA	- 0.055***	- 0.043***	- 0.057***	- 0.062***	- 0.031***	- 0.020***	
	(0.004)	(0.004)	(0.005)	(0.006)	(0.007)	(0.004)	
Non-EEA	- 0.031***	- 0.017***	-0.008	-0.005	- 0.058***	- 0.031***	
	(0.004)	(0.004)	(0.006)	(0.007)	(0.007)	(0.005)	
In couple	0.062***	0.015***	0.058***	0.015***	0.058***	0.011***	
	(0.003)	(0.002)	(0.003)	(0.004)	(0.004)	(0.003)	
With dependent chil- dren	0.026***	- 0.032***	0.023***	- 0.035***	0.025***	- 0.032***	
(yrs<18)							
	(0.003)	(0.003)	(0.004)	(0.005)	(0.005)	(0.003)	
In couple with dep. children	0.011**	0.021***	0.014**	0.041***	0.008	0.012**	
	(0.004)	(0.003)	(0.005)	(0.006)	(0.007)	(0.004)	
Workplace characteristics							
Part-time	- 0.097***	- 0.037***	- 0.083***	- 0.034***	- 0.095***	- 0.036***	
	(0.003)	(0.002)	(0.005)	(0.003)	(0.004)	(0.002)	

Table 6Mincerian regression results, years 2005–2020

Table 6	continued

	Dep. var.: L	og(Wage)					
	All sectors		Male-domin	nated sectors	Female-dominated sectors		
	Man (1)	Women (2)	Man (3)	Women (4)	Man (5)	Women (6)	
Public sector	0.031***	0.059***	0.003	0.005	0.059***	0.088***	
	(0.003)	(0.002)	(0.005)	(0.006)	(0.005)	(0.003)	
Low gender segregation	0.027***	-0.005	-0.141^{***}	-0.102^{***}	-0.006	-0.012^{***}	
	(0.003)	(0.003)	(0.009)	(0.013)	(0.006)	(0.003)	
Female dominance	- 0.163***	- 0.158***					
	(0.003)	(0.002)					
Female Dominance × Low gender segrega- tion	0.006	0.016***					
	(0.005)	(0.004)					
Working region controls	Yes	Yes	Yes	Yes	Yes	Yes	
Other socio- demographic controls	Yes	Yes	Yes	Yes	Yes	Yes	
SOC dummies	Yes	Yes	Yes	Yes	Yes	Yes	
SIC dummies	No	No	Yes	Yes	Yes	Yes	
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	
Observations	219,027	219,297	147,953	76,699	71,074	142,598	

Data from UK Labour Force Survey (LFS). Models (1)–(4) are estimated using OLS. Robust errors are in parenthesis. Significance levels: p < 0.01 ***, p < 0.05 **, p < 0.1 *

For socio-demographic and job characteristics, being non-UK natives is significantly associated with lower wages. However, the reduction is, on average, larger in absolute terms for EEA than non-EEA, except for female-dominated sectors. The presence of dependent children penalises women's wages but not men's, independently of the sector. Further, working in the public rather than in the private sector is associated with higher wages for women than men. However, the coefficients are non-significant in male-dominated sectors. This suggests that the private sector pays more in male-dominated sectors while the public sector offers better remuneration in female-dominated sectors.

5.2.3 Results based on predicted and residual wages

This section discusses empirical evidence on the differences in the selection of workers in male- and female-dominated sectors in terms of observable (predicted wages) and unobservable (residual wages) characteristics. Figures 4 and 5, respectively, display the CDFs of potential and residual wages for the four subgroups: men in male-dominated sectors (*ml*, *ml-dom*), women in male-dominated sectors (*fml*, *ml-dom*), men in female-dominated sectors (*ml*, *fml-dom*) and women in female-dominated sectors



Fig. 4 CDFs of predicted wages, by gender and sectoral dominance. Note: The solid line is for men

While a construction of predicted wages, by generating sectoral dominated. Note: The solid line is for menworking in male-dominated sectors, the short-dashed line is for women employed in male-dominated sectors, the long-dash line is for men in female-dominated sectors, and the dash-dot line is for women in female-dominated sectors. Left: Predicted wages are calculated after estimating the coefficients of the Mincerian wage regression, reported in Table 6. Right: predicted wages are calculates using the estimated coefficients from the Mincerian regression of men working in male-dominated sectors. Predicted wages in the counterfactual exercise are precise measure of individual earnings potential (Gould and Moav, 2016; Borjas et al., 2019)

tors (*finl*, *finl-dom*). The graphs on the left show actual values, calculated using the estimated coefficients for each subgroup from Table 6. The graphs on the right display counterfactual values calculated with the estimated coefficients of men working in male-dominated sectors.

The key result from the left graph in Fig.4 is that there is a penalty in potential wages associated with working in a female-dominated sector. In fact, women in female-dominated sectors always have lower predicted wages than all other workers (CDFs always lying on the left). For low levels of potential wages, men employed in female-dominated sectors earn much less than women in male-dominated sectors.

If workers had the potential wages of men in male-dominated sectors, wage differentials of men and women across female- and male-dominated sectors would be smaller (Fig. 4 on the right). However, women in female-dominated sectors would always be paid less than all other workers. Men in female-dominated sectors would still be penalised compared to workers in male-dominated sectors, but only in low-paid jobs. But moving to the top of the distribution, the gap in terms of potential counterfactual wages between women and men increases, meaning that women would always



Fig. 5 CDFs of residual wages, by gender and sectoral dominance. *Note*: The solid line is for men working in male-dominated sectors, the short-dashed line is for women employed in male-dominated sectors, the long-dash line is for men in female-dominated sectors, and the dash-dot line is for women in female-dominated sectors. Left: Residual wages are calculated after estimating the coefficients of the Mincerian wage regression, reported in Table 6. Right: Residual wages are calculates using the estimated coefficients from the Mincerian regression of men working in male-dominated sectors. Residuals from a Mincerian regression calculated in this way capture the part of earnings that is uncorrelated to observed skills (Parey et al., 2017)

earn less than men. These findings contrast Roos (1981) and Johnson and Solon (1984), who always find a more pronounced wage differential for men than women in femaledominated environments.

When we look at the residual wages in Fig. 5, the results highlight that differences in wages in high-paid jobs cannot be attributed to acquired skills or accumulated human capital only. Specifically, the CDF of women in female-dominated sectors (left graph) lies to the right of the other curves for low residual wages (positive selection at the bottom of the distribution) and to their left for high values (negative selection at the top). This means that these women earn more in low-paid jobs but much less in high-paid jobs than the other workers for reasons other than their skills and human capital.

In the counterfactual exercise (right graph), all curves would shift to the left of the CDF of male workers in male-dominated sectors, meaning that all workers would be negatively selected with respect to the former. At the bottom of the distribution, both women and men in female-dominated sectors would be penalised in terms of residual wages due to unobserved characteristics (the two CDFs overlap and lie to the left of

the other two). However, as we move up to the distribution, the CDFs diverge, and women in female-dominated sectors are more negatively selected (laying more to the left) than their male counterparts and other subgroups.

The non-parametric K–S test always rejects the null hypothesis of equality of distributions among the four sub-groups (see Table A7 in the Appendix), confirming that the distributions of (actual and counterfactual) predicted and residual wages of men and women across sectors differ.

6 Conclusion

This paper studied how gender sectoral segregation relates to employment contracts (i.e. permanent jobs, part-time jobs, working hours, remote work) and hourly wages in the UK between 2005 and 2020. We further analysed the extent to which wages differ in female- and male-dominated sectors by looking at both observable and unobservable characteristics. Our empirical analysis suggested that the persistent imbalance in the shares of men or women in some sectors contributes to explaining the differences in employment contracts and wages.

We first found that female-dominated sectors reflect contractual characteristics typical of women. In other words, working in female-dominated sectors is associated with a greater reliance on part-time contracts, fewer hours and less working from home, even controlling for the occupational composition. In addition, female-dominated sectors seem to pay, in general, less for any worker regardless of their gender. The penalty for men working in female-dominated sectors is even larger than for women.

Second, women working in female- and male-dominated sectors are paid less not because of differences in human capital and productivity but rather because of the existence of persistent "discriminatory constraints", such as barriers in the labour market for women due to the effects of discrimination and unobserved differences in productivity and tastes. This means that women have observable attributes similar to men regarding accumulated human capital, and without these discriminatory barriers, wage differentials between women and men within male- and female-dominated sectors would be lower.

Third, female-dominated sectors are not as rewarding as male-dominated sectors in terms of predicted and residual wages. While women in female-dominated sectors are always worse off than all other workers, men in female-dominated sectors are disadvantaged in low-paid jobs only. In addition, actual and counterfactual results for residual wages have documented the negative selection of women in female-dominated sectors with respect to all other workers, especially at the top of the wage distribution. The use of predicted and residual wages allows us to control the issue of selection based on unobservables that arises from the use of the PSM and Mincer regression, where the former matches workers with similar observed characteristics and the latter holds constant observed individual factors associated with wages.

This analysis has policy implications. Gender segregation in the labour market may be responsible for causing more challenges for women than their male counterparts regarding labour participation, access to jobs and career opportunities. This gap could potentially widen post-pandemic. Our findings can provide policy-makers with empirical evidence supporting appropriate reforms favouring vulnerable categories of workers (i.e. women, mothers and immigrants) and policies designed to sustain longrun economic growth, especially as the UK is facing new challenges (i.e. pandemic and Brexit). Future avenues of research could focus on gender segregation into sectors that have seen a rise in the use of atypical work arrangements, e.g. zero-hour contracts and casual work.

Funding Open access funding provided by Alma Mater Studiorum - Università di Bologna within the CRUI-CARE Agreement.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

Appendix A

See Tables A1, A2, A3, A4, A5, A6 and A7.

Occupations	1	2	3	4	5	6	7	8	9	Total
Sectors										
A—Agriculture, forestry and fishing	0.14	0.03	0.03	0.02	0.28	0.01	0.01	0.10	0.37	0.99
B—Mining and quarrying	0.13	0.17	0.11	0.02	0.13	0.00	0.01	0.21	0.02	0.81
C—Manufacturing	3.03	2.42	2.13	0.53	4.74	0.02	0.28	4.25	1.61	19.01
D-Electricity, gas and air con supply	0.14	0.24	0.15	0.04	0.27	0.00	0.09	0.08	0.03	1.04
E—Water supply, sewerage and waste	0.17	0.14	0.14	0.04	0.11	0.01	0.03	0.44	0.37	1.45
F—Construction	1.59	1.25	0.64	0.18	3.96	0.02	0.07	1.15	0.95	9.82
G—Distribution	2.87	0.54	1.15	0.65	1.94	0.02	3.90	1.69	2.40	15.17
H—Transport	0.78	0.29	0.63	0.43	0.42	0.23	0.16	3.20	1.71	7.85
I—Accommodation and food services	0.70	0.04	0.09	0.13	1.24	0.07	0.20	0.18	2.05	4.70
J-Information, communication	0.83	2.08	0.83	0.14	0.36	0.00	0.21	0.13	0.45	5.04
K—Financial and insurance services	1.26	0.92	1.47	0.68	0.04	0.00	0.28	0.02	0.06	4.74
L—Real estate services	0.32	0.13	0.25	0.08	0.11	0.05	0.05	0.02	0.05	1.06
M—Professional, scientific and technical activities	1.33	2.63	1.46	0.49	0.33	0.04	0.14	0.19	0.50	7.11
N—Admin and support services	0.75	0.60	0.61	0.21	0.48	0.16	0.28	0.25	1.10	4.42
P—Education	0.29	3.96	0.75	0.20	0.17	0.59	0.02	0.07	0.21	6.26
Q-Health and social work	0.66	1.85	1.08	0.35	0.24	1.49	0.06	0.15	0.44	6.32
R-Arts, entertainment and recreation	0.45	0.19	0.59	0.18	0.35	0.28	0.08	0.05	0.34	2.51
S—Other service activities	0.22	0.43	0.21	0.11	0.19	0.21	0.03	0.08	0.14	1.62
T—Households as employers	0.01	0.00	0.00	0.00	0.04	0.02	0.00	0.00	0.01	0.08
Total	15.68	17.90	12.33	4.48	15.42	3.22	5.89	12.25	12.83	100.00

Table A1 Share of workers by sector and occupation, male sample

The percentages are calculated on total male employed workers (self-employed are excluded). Labels for occupations: 1—Managers, Directors And Senior Official; 2—Professional Occ.; 3—Associate Professional And Technical Occ.; 4—Administrative And Secretarial Occ.; 5—Skilled Trades Occ.; 6—Caring, Leisure And Other Service Occ.; 7—Sales And Customer Service Occ.; 8—Process, Plant And Machine Operatives; 9—Elementary Occ

Table A2 Share of workers by sector and occupation, female sample

•					•					
Occupations	1	2	3	4	5	6	7	8	9	Total
Sectors										
A—Agriculture, forestry and fishing	0.04	0.02	0.02	0.09	0.04	0.05	0.02	0.02	0.11	0.41
B—Mining and quarrying	0.03	0.03	0.04	0.06	0.00	0.00	0.00	0.00	0.00	0.17
C—Manufacturing	0.79	0.52	0.98	1.61	0.27	0.04	0.34	1.17	0.66	6.38
D-Electricity, gas and air con supply	0.04	0.05	0.07	0.10	0.00	0.00	0.11	0.00	0.01	0.38
E-Water supply, sewerage and waste	0.04	0.04	0.06	0.12	0.00	0.00	0.04	0.01	0.03	0.35
F—Construction	0.27	0.19	0.26	0.98	0.04	0.01	0.09	0.01	0.06	1.91
G—Distribution	1.51	0.34	1.05	2.00	0.25	0.07	8.71	0.25	1.41	15.59
H—Transport	0.25	0.09	0.21	0.67	0.01	0.41	0.20	0.17	0.43	2.44
I—Accommodation and food services	0.58	0.03	0.16	0.41	0.49	0.27	0.40	0.03	3.70	6.09
J-Information, communication	0.31	0.50	0.47	0.40	0.02	0.01	0.27	0.01	0.12	2.11
K-Financial and insurance services	0.70	0.44	1.05	1.95	0.00	0.01	0.51	0.01	0.05	4.72
L—Real estate services	0.26	0.07	0.40	0.45	0.01	0.05	0.11	0.00	0.04	1.39
M—Professional, scientific and technical activities	0.78	1.38	1.29	2.02	0.03	0.18	0.22	0.04	0.35	6.28
N-Admin and support services	0.36	0.21	0.47	0.67	0.03	0.35	0.35	0.03	0.95	3.41
P—Education	0.39	8.30	1.14	1.71	0.20	4.69	0.06	0.01	1.76	18.27
Q-Health and social work	1.60	5.86	4.17	3.16	0.20	8.65	0.20	0.04	1.00	24.87
R-Arts, entertainment and recreation	0.31	0.21	0.44	0.61	0.04	0.28	0.15	0.01	0.39	2.43
S—Other service activities	0.20	0.25	0.25	0.44	0.03	1.07	0.09	0.02	0.22	2.58
T—Households as employers	0.00	0.00	0.00	0.01	0.00	0.17	0.00	0.00	0.04	0.22
Total	8.46	18.51	12.54	17.46	1.68	16.31	11.86	1.85	11.34	100.00

The percentages are calculated on total female employed workers (self-employed are excluded). Labels for occupations: 1—Managers, Directors And Senior Official; 2—Professional Occ.; 3—Associate Professional And Technical Occ.; 4—Administrative And Secretarial Occ.; 5—Skilled Trades Occ.; 6—Caring, Leisure And Other Service Occ.; 7—Sales And Customer Service Occ.; 8—Process, Plant And Machine Operatives; 9—Elementary Occ

	Log(wage) (1)	Permanent job (2)	Part-time job (3)	Log(hours) (4)	Remote work (5)	Female-dom. sector (6)
Female	- 0.006		0.048	- 0.081		0.095
In couple		0.008			0.0112	
With dep. Children (yrs<18)	0.004				0.001	
Female x In couple						
01	0.075	0.003	- 0.009	0.025	0.000	- 0.027
11			0.143	-0.122		0.019
10		0.003				
Female x Dep. Children						
01	0.035		0.045	0.013		
10	- 0.049				- 0.013	0.027
11		0.002		- 0.096		
In couple x Dep. Children						
01			0.003	- 0.033	0.001	0.024
10	-0.011					0.001
11	0.017					
Age	0.012	0.004	- 0.030	0.039	- 0.009	-0.022
Age Sq		-0.000	0.000	-0.000	0.000	0.000
EEA	-0.026	-0.022	-0.077	0.093	-0.002	0.007
non-EEA	- 0.035	-0.020	-0.014	0.024		0.034
Black	- 0.059	- 0.019	-0.037	0.031	-0.012	0.025
Asian	- 0.066	-0.005	-0.025	0.001	-0.014	0.004
Other ethnicity	-0.066	- 0.013	-0.004	-0.005	-0.012	0.047
Muslim	- 0.097	0.010	0.124	- 0.122	-0.000	0.099
Christian	- 0.049	0.001	0.002	-0.000	-0.005	0.002
Other religion		-0.004	-0.005	0.018	-0.006	0.007
Intermediate Education	0.033	-0.000	0.011	-0.015	0.006	- 0.011
Higher Education	0.109		0.029	- 0.039	0.013	-0.048
Years of Education	0.168	- 0.016	-0.014	0.008	0.0164	-0.021
Years of Education Sq	-0.005	0.000	0.000	0.000	-0.000	0.001
Experience	0.013	0.004	0.005	-0.008	0.0060	0.001
Experience Sq	-0.000	-0.000	-0.000	0.000	-0.000	-0.000
Training Opportunity	0.002	0.020	-0.051	0.0534	-0.012	- 0.012
Public Sector	- 0.021	- 0.029	0.037	-0.062	-0.062	0.379
SOC	Yes	Yes	Yes	Yes	Yes	Yes
Female x SOC	Yes	Yes	Yes	Yes	Yes	Yes
Working Regions	Yes	Yes	Yes	Yes	Yes	Yes
$\lambda(BIC)$	120.7	36.5	46.9	42.6	23.3	63.4

Table A3 Post-estimation OLS coefficients of variables selected by LASSO

The table reports the post-estimation OLS coefficients of the variables selected by Lasso. The estimated models correspond to those with minimum BIC. The STATA command used to obtain the estimates is lasso2

	Full sample	Male sample	Female sample
Dep. var: Female-dominated sec	ctors		
Woman	1.063***		
	(0.007)		
Woman not in couple	-0.218***		
	(0.007)		
Woman w/t dep. children	-0.059***		
	(0.010)		
In couple	-0.108***	-0.060^{***}	0.086***
	(0.006)	(0.007)	(0.007)
Dep. children	0.120***	0.028**	0.252***
	(0.007)	(0.009)	(0.008)
In couple w/dep. children	- 0.043***	0.007	-0.075***
	(0.012)	(0.017)	(0.016)
EEA	-0.044***	0.115***	-0.189***
	(0.008)	(0.012)	(0.011)
Non-EEA	0.074***	0.158***	-0.035**
	(0.009)	(0.012)	(0.012)
Age	- 0.033***	-0.041^{***}	-0.036***
	(0.001)	(0.002)	(0.002)
Age squared	0.000***	0.001***	0.001***
	(0.000)	(0.000)	(0.000)
Higher educ	0.004	0.057***	-0.045^{***}
	(0.005)	(0.007)	(0.007)
Years of educ	0.041***	0.060***	0.061***
	(0.007)	(0.009)	(0.009)
Years of educ. sqr	-0.001**	-0.002^{***}	-0.001
	(0.000)	(0.000)	(0.000)
Black	0.175***	0.169***	0.192***
	(0.014)	(0.020)	(0.020)

Table A4 Probit for female-dominated sectors

	Full sample	Male sample	Female sample
Asian	0.007	0.070***	- 0.058***
	(0.012)	(0.017)	(0.017)
Other ethnicity	0.171***	0.255***	0.091***
	(0.013)	(0.017)	(0.017)
Muslim	0.326***	0.279***	0.219***
	(0.016)	(0.019)	(0.025)
Christian	0.021***	-0.019**	0.059***
	(0.004)	(0.006)	(0.006)
Other religion	0.036***	0.023*	0.040***
	(0.007)	(0.009)	(0.010)
Region Controls	Yes	Yes	Yes
SOC	Yes	Yes	Yes
Male x SOC	Yes	Yes	Yes
Observations	617,119	296,465	320,654

Table A4 continued

The estimates are used to calculate propensity scores for the PSM. Robust standard errors. Significance levels: p value < 0.01 ***, p value < 0.05 **, p value < 0.1 *

ors
d sect
minated
ale-do
, fem
decomposition
30
K
of
omponents
alc
idu
indiv
of
ibution
ontr

Table A5 Cor	tribution o	f individua	l compone.	nts of KBC) decompos	sition, fen	aale-domin	lated sector	S						
Year	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020
Endowment ej	Fect														
Low gender sect.	0.01^{***}	0.01^{**}	0.01***	0.02***	0.01^{**}	0.01^{***}	0.02***	0.02***	0.01^{***}	0.01^{***}	0.00	0.01^{***}	0.01^{***}	0.01^{***}	0.03***
segregation															
	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.004)	(0.004)	(0.003)	(0.003)	(0.003)	(0.004)	(0.004)	(0.004)	(0.005)
Log(hours)	-0.06^{**}	$^{*}-0.02$	- 0.08***	*-0.05***	*-0.05***	- 0.03**	$^{*}-0.04^{***}$	*-0.03**	-0.04^{***}	*-0.03**	-0.06^{***}	*- 0.03**	-0.04^{***}	*- 0.04**	-0.04^{**}
	(0.014)	(0.014)	(0.015)	(0.013)	(0.013)	(0.013)	(0.014)	(0.013)	(0.013)	(0.015)	(0.014)	(0.013)	(0.015)	(0.017)	(0.015)
Log(hours) ²	0.06^{***}	0.01	0.08^{***}	0.05***	0.05***	0.03*	0.04^{**}	0.03^{**}	0.04^{***}	0.03*	0.06***	0.03*	0.04^{**}	0.03*	0.03*
	(0.016)	(0.015)	(0.016)	(0.015)	(0.015)	(0.015)	(0.015)	(0.015)	(0.015)	(0.016)	(0.015)	(0.015)	(0.017)	(0.018)	(0.017)
EEA	-0.00	-0.00	0.00	0.00	0.00	-0.00	0.00	-0.00	0.00	-0.00	-0.00	-0.00	-0.00	-0.00	-0.00
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.00)
Non-EEA	-0.00*	-0.00*	-0.00	-0.00	-0.00	-0.00	-0.00	-0.00	-0.00	-0.00**	*- 0.00***	*- 0.00**	-0.00	-0.00	-0.00
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.00)
Age	-0.00	0.00	0.00	-0.01^{**}	-0.00	-0.00	-0.01*	0.01	0.00	0.00	0.01	0.00	-0.00	0.01	-0.02^{**}
	(0.002)	(0.001)	(0.005)	(0.006)	(0.005)	(0.005)	(0.007)	(0.006)	(0.006)	(0.006)	(0.006)	(0.005)	(0.007)	(0.006)	(0.008)
Age ²	0.00	0.00	0.00	0.03^{***}	-0.00	-0.01*	0.00	-0.02^{**}	-0.01^{**}	-0.01^{*}	-0.01^{**}	-0.01	-0.01	-0.01^{**}	0.00
	(0.003)	(0.004)	(0.008)	(0.00)	(0.004)	(0.005)	(0.005)	(0.007)	(0.006)	(0.005)	(0.006)	(0.005)	(0.006)	(0.006)	(0.006)
Experience	-0.00	-0.00	0.00	0.01^{**}	-0.01^{**}	-0.00	-0.00	-0.01^{***}	*- 0.02***	*- 0.02***	*- 0.02***	*- 0.02***	- 0.02***	*- 0.02***	-0.00
	(0.002)	(0.003)	(0.003)	(0.005)	(0.006)	(0.005)	(0.005)	(0.005)	(0.006)	(0.006)	(0.006)	(0.006)	(0.006)	(0.006)	(0.004)
Experience ²	-0.00	-0.01^{***}	*- 0.02***	*-0.03***	*0.01**	0.02^{***}	0.01^{***}	0.02^{***}	0.02^{***}	0.02^{***}	0.03***	0.02^{***}	0.02^{***}	0.02^{***}	0.01^{**}
	(0.004)	(0.004)	(0.005)	(0.007)	(0.004)	(0.006)	(0.004)	(0.006)	(0.006)	(0.006)	(0.006)	(0.006)	(0.006)	(0.006)	(0.005)

Table A5 cont	tinued														
Year	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020
Intermediate educ	-0.00	$- 0.00^{**}$	$- 0.00^{**}$	-0.00^{**}	$- 0.00^{*}$	0.00*	-0.00	-0.00	-0.00	-0.00*	0.00	-0.00	0.00	- 0.00	0.00
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
High educ	0.00^{**}	0.00^{***}	0.01^{***}	0.00^{***}	0.00***	0.00*	0.00***	0.00^{**}	0.00^{**}	0.00^{***}	0.00*	0.00^{**}	0.00	0.00*	0.00
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)
Years educ	0.00	-0.01	-0.01*	-0.01^{**}	-0.00	-0.00	-0.02^{**}	-0.01^{**}	-0.02^{**}	-0.02^{**}	*-0.01**	-0.03^{**}	*- 0.02**	*- 0.02**:	*- 0.03**
	(0.004)	(0.004)	(0.006)	(0.006)	(0.005)	(0.006)	(0.007)	(0.007)	(0.006)	(0.005)	(0.006)	(0.007)	(0.008)	(0.006)	(0.008)
Years educ ²	0.00	0.01^{**}	0.01^{***}	0.01^{***}	0.01	0.01	0.02^{***}	0.02^{***}	0.02^{***}	0.02^{***}	0.01^{**}	0.03^{***}	0.02^{***}	0.02^{***}	0.03^{***}
	(0.003)	(0.004)	(0.005)	(0.005)	(0.005)	(0.006)	(0.006)	(0.006)	(0.006)	(0.005)	(0.005)	(0.006)	(0.007)	(0.006)	(0.008)
Training	-0.00**	*-0.00***	*- 0.00***	·- 0.00***	-0.00^{***}	·- 0.00**·	*-0.00**	-0.00^{**}	*-0.00*	-0.00	-0.00	-0.00	-0.00	-0.00	-0.00
	(0.001)	(0.001)	(0.001)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
In couple	0.00	-0.00	-0.00	0.00	0.00	-0.00	-0.00	-0.00	-0.00	0.00	0.00	0.00	-0.00	0.00	-0.00
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
With dep. children	0.00***	0.01***	0.01***	0.01^{***}	0.00***	0.00***	0.00***	0.00***	0.00***	0.00***	0.00***	0.00***	0.00***	0.00**	0.00
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.000)
In couple with dep. childrer	1 - 0.00	- 0.00	- 0.00	- 0.00*	- 0.00	0.00	- 0.00	0.00	-0.00	0.00	0.00	0.00	0.00	0.00	0.00
	(0.000)	(0.001)	(0.000)	(0.001)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Public sector	-0.02^{**}	*- 0.02***	*- 0.02***	- 0.02***	-0.02^{***}	·- 0.02**·	*- 0.02***	*- 0.02**	*- 0.01**	*- 0.01**	*- 0.01**	$*-0.01^{**}$	*- 0.01**	*- 0.01**:	-0.01**
	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)

 $\underline{\textcircled{O}}$ Springer

Table A5 con	ntinued														
Year	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020
Coefficient eff	fect														
Low gender sect.	0.02^{**}	0.03***	*0.03***	0.01	0.04^{***}	0.02*	0.02^{**}	0.02^{**}	0.02***	0.03***	0.03***	0.01	0.02**	0.03***	0.00
segregation															
	(0.00)	(0.00)	(600.0)	(0.008)	(600.0)	(0.008)	(0.008)	(0.008)	(0.008)	(0.008)	(0.008)	(0.008)	(0.008)	(0.009)	(0.010)
Log(hours)	0.73^{**}	-0.36	1.05^{***}	1.78^{***}	-0.26	1.47^{***}	1.00^{***}	1.16^{***}	-0.39	0.53	1.28^{***}	1.38^{***}	1.05^{***}	1.23^{***}	1.51^{***}
	(0.326)	(0.339)	(0.320)	(0.337)	(0.279)	(0.310)	(0.304)	(0.364)	(0.290)	(0.323)	(0.396)	(0.386)	(0.399)	(0.394)	(0.467)
Log(hours) ²	-0.44^{**}	**0.10	-0.59^{**}	*-1.00**	*0.06	- 0.73**:	*- 0.57**:	*- 0.59**:	*0.19	-0.24	$-0.65^{**:}$	*- 0.71***	*-0.51**	*- 0.68**:	*- 0.75***
	(0.169)	(0.179)	(0.172)	(0.181)	(0.153)	(0.166)	(0.164)	(0.191)	(0.157)	(0.173)	(0.208)	(0.202)	(0.209)	(0.211)	(0.252)
EEA	0.00	0.00	0.00	-0.00	-0.00	0.00	-0.00	-0.00	-0.00**	*-0.00**	*-0.00	0.00	0.00	0.00	-0.00
	(0.001)	(0.001)	(0.001)	(0.001)	(0.002)	(0.001)	(0.001)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)
Non-EEA	0.00	0.00	-0.00	-0.00	-0.00	-0.00^{**}	-0.00*	0.00	-0.00	-0.00	0.00	-0.00	-0.01^{**}	$^{k} - 0.00$	0.00
	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.003)	(0.003)	(0.002)	(0.003)	(0.003)	(0.003)	(0.003)
Age	0.02	0.76*	0.53	0.70*	0.45	0.70	0.02	1.30^{***}	0.58	0.03	0.10	0.66	-0.47	0.40	-0.38
	(0.415)	(0.402)	(0.415)	(0.412)	(0.467)	(0.437)	(0.423)	(0.454)	(0.435)	(0.459)	(0.422)	(0.455)	(0.451)	(0.487)	(0.577)
Age^{2}	-0.45^{**}	^k - 0.37 ^s	*-0.19	-0.18	0.06	$-0.68^{**:}$	*0.07	$-0.76^{**:}$	*-0.32	0.12	-0.01	-0.06	0.20	0.01	0.21
	(0.198)	(0.197)	(0.196)	(0.198)	(0.229)	(0.216)	(0.218)	(0.225)	(0.222)	(0.236)	(0.222)	(0.231)	(0.239)	(0.254)	(0.298)
Experience	0.47***	-0.03	-0.02	-0.16	-0.02	0.22	0.19	-0.13	0.02	0.06	-0.00	-0.24	0.18	-0.09	0.23
	(0.163)	(0.153)	(0.161)	(0.159)	(0.184)	(0.169)	(0.160)	(0.168)	(0.159)	(0.170)	(0.146)	(0.162)	(0.154)	(0.163)	(0.206)
Experience ²	0.03	0.04	-0.03	-0.00	-0.17^{**}	*0.09	-0.14^{*}	0.15^{*}	0.00	-0.10	-0.02	-0.02	-0.06	-0.07	-0.16
	(0.068)	(0.067)	(0.067)	(0.067)	(0.080)	(0.075)	(0.075)	(0.077)	(0.077)	(0.080)	(0.075)	(0.078)	(0.080)	(0.085)	(0.103)

Table A5 conti	nued														
Year	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020
Intermediate educ	0.02***	0.00	0.00	-0.01	0.00	0.02***	-0.01	0.00	0.00	0.00	-0.00	- 0.00	0.01	-0.01	- 0.01
	(0.006)	(0.006)	(0.006)	(0.007)	(0.007)	(0.007)	(0.007)	(0.007)	(0.007)	(0.008)	(0.007)	(0.008)	(0.008)	(0.008)	(0.010)
High educ	0.03***	0.00	-0.01	-0.01	-0.01	0.01	-0.01	0.00	-0.00	-0.01	-0.00	-0.01	0.01	-0.02	-0.01
	(0.007)	(0.008)	(0.008)	(600.0)	(0.010)	(0.00)	(0.010)	(0.010)	(0.010)	(0.011)	(0.011)	(0.011)	(0.012)	(0.012)	(0.016)
Years educ	0.43	0.45	0.23	0.39	0.19	0.02	0.10	-0.26	0.31	0.39	0.62	0.56	-0.03	0.46	-0.14
	(0.299)	(0.313)	(0.321)	(0.325)	(0.360)	(0.345)	(0.360)	(0.365)	(0.362)	(0.386)	(0.384)	(0.396)	(0.401)	(0.440)	(0.490)
Years educ ²	-0.22	-0.23	-0.15	-0.21	-0.11	-0.03	-0.04	0.13	-0.14	-0.20	-0.30*	-0.28	0.02	-0.22	0.10
	(0.137)	(0.144)	(0.149)	(0.151)	(0.168)	(0.161)	(0.170)	(0.173)	(0.171)	(0.184)	(0.183)	(0.190)	(0.192)	(0.212)	(0.238)
Training	0.02^{***}	0.00	0.01	0.03^{***}	0.00	-0.00	0.00	0.00	0.01	-0.00	0.01^{*}	0.00	0.00	0.00	0.01^{*}
	(0.008)	(0.008)	(0.009)	(0.00)	(0.00)	(0.004)	(0.004)	(0.004)	(0.003)	(0.004)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)
In couple	0.04^{***}	0.02^{**}	0.04^{***}	0.02	0.02	0.00	0.03^{***}	0.03^{***}	0.04^{***}	0.06^{***}	0.05^{***}	0.03^{***}	0.01	0.02^{*}	0.04^{***}
	(0.012)	(0.012)	(0.011)	(0.011)	(0.013)	(0.012)	(0.012)	(0.012)	(0.011)	(0.012)	(0.012)	(0.012)	(0.012)	(0.013)	(0.012)
With dep. children	0.01	0.04***	0.03***	0.03**	0.04***	0.04***	0.04***	0.02^{*}	0.04***	0.03^{***}	0.04***	0.04^{***}	0.04^{***}	0.04^{***}	0.00
	(0.011)	(0.012)	(0.012)	(0.012)	(0.012)	(0.011)	(0.011)	(0.012)	(0.011)	(0.011)	(0.011)	(0.012)	(0.012)	(0.012)	(0.010)
In couple with dep. children	0.01	0.00	0.00	0.01	-0.00	-0.00	-0.01	0.01	-0.01	-0.01	-0.02^{*}	- 0.01	-0.00	-0.01	0.01
	(0.00)	(0.010)	(0.010)	(0.010)	(0.010)	(0.009)	(600.0)	(0.009)	(0.009)	(0.00)	(600.0)	(0.00)	(0.00)	(0.010)	(0.008)
Public sector	-0.01	-0.02^{**}	-0.01	-0.02*	-0.01	0.01	-0.03^{***}	- 0.02*	-0.01	0.01	-0.01	0.00	-0.00	-0.02*	-0.01
	(0.008)	(0.009)	(0.009)	(0.010)	(0.010)	(0.00)	(600.0)	(0.009)	(0.00)	(600.0)	(600.0)	(0.00)	(600.0)	(0.010)	(0.012)
Constant	-0.51^{*}	-0.25	-0.80^{***}	-1.21^{***}	* - 0.14	-0.98^{***}	-0.61^{**}	-1.01^{***}	-0.23	-0.59^{**}	-1.05^{***}	- 1.27***	- 0.42	-1.01^{***}	-0.59
	(0.277)	(0.278)	(0.278)	(0.283)	(0.286)	(0.282)	(0.283)	(0.307)	(0.283)	(0.300)	(0.315)	(0.325)	(0.322)	(0.345)	(0.393)

Table A5 cont	tinued														
Year	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020
Interaction eff Low gender sect. segregation	fe <i>ct</i> 0.01**	0.02***	• 0.02***	0.01	0.03***	• 0.01*	0.01**	0.01**	0.01***	0.02***	0.02***	0.01	0.01**	0.02***	0.00
))	(0.005)	(0.005)	(0.005)	(0.006)	(0.006)	(0.006)	(0.005)	(0.006)	(0.005)	(0.005)	(0.006)	(0.006)	(0.006)	(0.006)	(0.008)
Log(hours)	0.08^{**}	-0.04	0.11^{***}	0.17^{***}	-0.03	0.15^{***}	0.10^{***}	0.12^{***}	-0.04	0.05	0.11^{***}	0.12^{***}	0.09***	0.10^{***}	0.11^{***}
	(0.038)	(0.037)	(0.034)	(0.033)	(0.028)	(0.032)	(0.030)	(0.037)	(0.026)	(0.029)	(0.035)	(0.035)	(0.035)	(0.032)	(0.034)
Log(hours) ²	-0.10^{**}	* 0.02	-0.12^{***}	* - 0.19 * *	* 0.01	-0.15^{***}	$* - 0.11^{***}$	$* - 0.11^{***}$	* 0.03	- 0.04	-0.11^{***}	-0.12^{***}	-0.09**	-0.11^{***}	-0.10^{***}
	(0.038)	(0.038)	(0.035)	(0.034)	(0.030)	(0.033)	(0.031)	(0.037)	(0.028)	(0.030)	(0.036)	(0.035)	(0.036)	(0.033)	(0.035)
EEA	0.00	0.00	0.00	-0.00	-0.00	0.00	-0.00	-0.00	-0.00	-0.00	-0.00	0.00	0.00	0.00	-0.00
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.00)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.00)	(0.000)
Non-EEA	0.00	0.00	-0.00	-0.00	-0.00	-0.00^{**}	-0.00*	0.00	-0.00	-0.00	0.00	-0.00	-0.00*	-0.00	0.00
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.002)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.000)
Age	-0.00	0.00	0.01	0.01	-0.01	-0.01	-0.00	-0.03^{**}	-0.01	-0.00	-0.00	-0.01	0.01	-0.01	0.01
	(0.002)	(0.004)	(600.0)	(0.00)	(0.010)	(0.00)	(0.010)	(0.012)	(0.011)	(0.010)	(0.010)	(600.0)	(0.012)	(0.011)	(0.011)
Age^{2}	-0.01	-0.01	-0.01	-0.01	-0.00	0.02*	-0.00	0.03^{**}	0.01	-0.00	0.00	0.00	-0.01	-0.00	-0.01
	(0.005)	(0.007)	(0.013)	(0.012)	(0.006)	(0.008)	(0.008)	(0.011)	(0.009)	(0.008)	(600.0)	(0.007)	(0.011)	(0.010)	(0.010)
Experience	-0.01^{*}	0.00	-0.00	-0.00	0.00	-0.01	-0.01	0.01	-0.00	-0.00	0.00	0.01	-0.01	0.00	-0.01
	(0.006)	(0.001)	(0.003)	(0.004)	(0.00)	(0.008)	(600.0)	(0.008)	(0.009)	(0.008)	(0.008)	(0.008)	(0.008)	(0.008)	(0.008)
Experience ²	0.00	0.00	-0.00	-0.00	0.01^{*}	-0.00	0.01^{*}	-0.01*	-0.00	0.01	0.00	0.00	0.00	0.01	0.01
	(0.001)	(0.004)	(0.007)	(0.007)	(0.005)	(0.004)	(0.005)	(0.005)	(0.005)	(0.005)	(0.006)	(0.005)	(0.006)	(0.006)	(0.007)

Table A5 con	tinued														
Year	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020
Intermediate educ	-0.00^{**}	* - 0.00	-0.00	0.00	-0.00	-0.00^{**}	0.00	- 0.00	- 0.00	-0.00	-0.00	0.00	-0.00	0.00	0.00
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.000)	(0.000)	(000.0)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.001)
High educ	0.01^{***}	0.00	-0.00	-0.00	-0.00	0.00	-0.00	0.00	-0.00	-0.00	-0.00	-0.00	0.00	-0.00	-0.00
	(0.002)	(0.002)	(0.002)	(0.001)	(0.002)	(0.002)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)
Years educ	0.00	-0.00	-0.00	-0.00	-0.00	-0.00	-0.00	0.00	-0.00	-0.01	-0.01	-0.01	0.00	-0.01	0.00
	(0.002)	(0.002)	(0.003)	(0.003)	(0.001)	(0.001)	(0.003)	(0.003)	(0.004)	(0.006)	(0.004)	(0.007)	(0.005)	(0.007)	(0.009)
Years educ ²	0.00	0.00	0.00	0.00	0.00	0.00	0.00	-0.00	0.00	0.01	0.01	0.01	-0.00	0.01	-0.00
	(0.002)	(0.003)	(0.004)	(0.004)	(0.002)	(0.002)	(0.004)	(0.004)	(0.004)	(0.006)	(0.004)	(0.007)	(0.006)	(0.007)	(0.009)
Training	-0.00*	$^{4}-0.00$	-0.00	$-0.00^{**:}$	* - 0.00	0.00	-0.00	-0.00	-0.00	0.00	-0.00	-0.00	-0.00	-0.00	-0.00
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.00)	(0.000)	(0.000)	(0.00)	(0.001)	(0.000)	(0.000)	(0.000)	(0.000)	(0.001)
In couple	-0.00**	-0.00*	-0.00	-0.00	-0.00	-0.00	-0.00	-0.00	-0.00	0.00	-0.00	0.00	-0.00	0.00	0.00
	(0.001)	(0.001)	(0.001)	(0.000)	(0.001)	(0.00)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.000)	(0.000)	(0.001)
With dep. children	- 0.00	-0.01^{**}	* - 0.01**	*-0.01**	-0.01^{***}	*- 0.00***	*- 0.00**	* - 0.00*	- 0.00**:	* - 0.00**	* - 0.00**:	* - 0.00**:	*- 0.00**:	* - 0.00**	- 0.00
	(0.002)	(0.003)	(0.002)	(0.002)	(0.002)	(0.002)	(0.001)	(0.002)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.000)
In couple with dep. childrei	n — 0.00 n	- 0.00	-0.00	-0.00	0.00	-0.00	-0.00	0.00	- 0.00	-0.00	-0.00	-0.00	-0.00	- 0.00	0.00
	(0.001)	(0.001)	(0.001)	(0.001)	(0.000)	(0.00)	(0.000)	(0.000)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)
Public sector	0.00	0.01^{**}	0.01	0.01^{*}	0.00	-0.00	0.01^{***}	0.01^{*}	0.01	-0.00	0.01	-0.00	0.00	0.01^{*}	0.00
	(0.003)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)
Observations	15,020	15,308	14,911	14,172	13,246	13,737	13,336	13,313	13,433	12,879	12,468	12,945	12,151	11,563	10,383
Contribution (levels: p value	of main so	cio-demog	raphic chai $< 0.05 **,$	racteristics, p value <	, human caj 0.1 *	pital attribu	tes and sec	ctoral indi	cators. Oc	cupations	(SOC) are	omitted for	layout rea	sons. Sign	ificance

Gender sectoral segregation in the UK

			•		•										
Year	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020
Endowment ej	ffect														
Low gender sect.	0.05^{***}	0.04***	0.05***	0.06***	0.05***	0.04***	0.05***	0.04^{***}	0.03***	0.03***	0.02**	0.02***	0.03***	0.04***	0.05***
segregation															
	(0.011)	(0.009)	(0.010)	(0.013)	(0.012)	(0.014)	(0.011)	(0.011)	(0.00)	(0.00)	(0.011)	(0.009)	(0.010)	(0.009)	(0.010)
Log(hours)	-0.18^{**}	*-0.19**	$^{*-} 0.21^{**}$	*-0.19**	*- 0.23**	*- 0.25**	*-0.19**	*- 0.12**	*- 0.23**	*- 0.28**:	*- 0.20***	$*-0.27^{***}$	·- 0.21**·	$*-0.11^{***}$	-0.12^{***}
	(0.020)	(0.020)	(0.019)	(0.022)	(0.025)	(0.022)	(0.021)	(0.024)	(0.027)	(0.021)	(0.024)	(0.026)	(0.024)	(0.022)	(0.025)
Log(hours) ²	0.18^{***}	0.19***	0.22^{***}	0.18^{***}	0.23^{***}	0.25***	0.19^{***}	0.12^{***}	0.23^{***}	0.28^{***}	0.20^{***}	0.27^{***}	0.22^{***}	0.10^{***}	0.12***
	(0.022)	(0.022)	(0.022)	(0.024)	(0.028)	(0.025)	(0.024)	(0.027)	(0.030)	(0.023)	(0.027)	(0.028)	(0.026)	(0.024)	(0.028)
EEA	-0.00	0.00	-0.00	-0.00	0.00	0.00	0.00*	0.00^{**}	0.00	0.00^{***}	0.00	0.00^{***}	0.00^{**}	0.00^{**}	0.00**
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)
Non-EEA	-0.00	-0.00	-0.00	-0.00	-0.00	0.00	-0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Age	0.01	0.12^{***}	0.08^{**}	0.08^{***}	0.02	-0.00	0.05^{**}	0.03	0.03*	0.05***	0.05***	-0.01	0.03*	0.05***	0.05**
	(0.032)	(0.031)	(0.032)	(0.030)	(0.023)	(0.018)	(0.022)	(0.022)	(0.020)	(0.019)	(0.018)	(0.015)	(0.015)	(0.018)	(0.024)
Age^2	0.01	-0.06^{**}	-0.00	-0.04	0.01	0.02	-0.02	-0.02	-0.03	-0.02	-0.04^{**}	0.01	-0.02	-0.04^{**}	-0.02
	(0.033)	(0.032)	(0.033)	(0.031)	(0.023)	(0.017)	(0.022)	(0.022)	(0.020)	(0.018)	(0.018)	(0.015)	(0.015)	(0.018)	(0.023)
Experience	0.07***	0.00	0.03	0.01	0.03*	0.05***	0.02	0.04^{***}	0.03^{**}	0.01	0.02	0.05***	0.02^{*}	0.02^{*}	0.01
	(0.022)	(0.021)	(0.022)	(0.020)	(0.016)	(0.014)	(0.016)	(0.015)	(0.015)	(0.013)	(0.012)	(0.013)	(0.011)	(0.012)	(0.016)
Experience ²	-0.09^{**}	$*-0.06^{**}$	$^{*-} 0.10^{**}$	*- 0.05**	*- 0.06**	*- 0.05**	*- 0.05**	$*-0.04^{**:}$	*- 0.03**	*- 0.04**:	*-0.02*	-0.04^{***}	- 0.02**	-0.02^{**}	-0.04^{**}
	(0.019)	(0.018)	(0.019)	(0.017)	(0.014)	(0.011)	(0.012)	(0.013)	(0.011)	(0.011)	(0.010)	(0.010)	(0.00)	(0.010)	(0.014)

Table A6 Contribution of individual components of KBO decomposition, male-dominated sectors

Table A6 conti	nued														
Year	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020
Intermediate educ	-0.00^{***}	-0.00*	-0.00	-0.00	$- 0.00^{**}$	- 0.00	-0.00	-0.01^{***}	-0.00	-0.00	-0.00	-0.00^{**}	$- 0.00^{**}$	- 0.00*	-0.00
	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)
High educ	0.00	-0.00*	-0.00	-0.00	-0.00	-0.00	-0.00	-0.00*	-0.00^{**}	-0.00	-0.00*	-0.01^{***}	-0.01^{***}	-0.01^{***}	-0.00*
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.002)	(0.001)	(0.001)	(0.001)	(0.002)	(0.002)	(0.002)	(0.002)
Years educ	0.03^{***}	0.02^{**}	0.02^{**}	0.00	0.02^{**}	0.02^{***}	0.01^{*}	-0.00	0.01	-0.00	0.00	-0.00	-0.01	-0.02^{**}	-0.01^{**}
	(0.007)	(0.007)	(0.007)	(0.007)	(0.007)	(0.008)	(0.006)	(0.008)	(0.006)	(0.007)	(0.006)	(0.005)	(0.007)	(0.008)	(0.007)
Years educ ²	-0.02^{***}	-0.02^{**}	-0.01*	-0.00	-0.01^{**}	-0.02^{**}	-0.01*	0.00	-0.01	0.00	-0.00	0.00	0.01	0.02^{**}	0.01^{*}
	(0.006)	(0.007)	(0.006)	(0.006)	(0.006)	(0.007)	(0.005)	(0.007)	(0.005)	(0.006)	(0.005)	(0.004)	(0.006)	(0.008)	(0.006)
Training	0.00^{**}	0.00*	0.00^{**}	0.00^{***}	0.00^{**}	0.00	-0.00	0.00	0.00	0.00	0.00	0.00	-0.00	-0.00	-0.00*
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.000)	(0.000)	(0.00)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.001)
In couple	-0.00	-0.00	-0.00^{**}	-0.00*	0.00	0.00	-0.00	0.00	-0.00	0.00	0.00*	0.00^{**}	0.00	0.00	0.00
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)
With dep. children	0.00**	0.00	0.00	**00.0	0.00	0.00	0.00	0.00*	0.00	0.00	0.00	0.00	0.00	0.00	- 0.00
	(0.001)	(0.001)	(0.00)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
In couple with dep. children	0.00**	0.00*	0.00	0.00	0.00	0.00	0.00*	0.00*	0.01^{***}	0.00	0.00	0.00	0.00	0.00	0.00
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.002)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)
Public sector	0.00	0.00	-0.00	-0.00	-0.00	-0.00	-0.00	-0.00	0.00	0.00	0.00	0.00	0.00	0.00^{**}	-0.00
	(0.000)	(0.000)	(0.000)	(0.000)	(0.001)	(0.000)	(0.000)	(0.000)	(0.000)	(0.00)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)

Gender sectoral segregation in the UK

Table A6 con	tinued														
Year	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020
Coefficient eff	ect 0.01	000	10.0	000	0.03	000	000	CO 0 -	- 0.03	- 0.03	V0 0	- 0.03	00.0	000	**900
sect.	10.0	00.00	10.0	00.00	c0.0	70.0	70.02	70.0				C0.0 -	00.0	70.02	0000
segregation															
	(0.023)	(0.020)	(0.022)	(0.026)	(0.028)	(0.029)	(0.024)	(0.023)	(0.022)	(0.023)	(0.028)	(0.023)	(0.025)	(0.024)	(0.028)
Log(hours)	2.98***	2.83***	3.10^{***}	3.71^{***}	1.95^{***}	1.93^{***}	1.67^{***}	1.25^{***}	1.84^{***}	4.47***	2.33***	3.00^{***}	2.46***	2.91***	1.16^{**}
	(0.395)	(0.361)	(0.368)	(0.392)	(0.427)	(0.342)	(0.402)	(0.388)	(0.435)	(0.458)	(0.396)	(0.427)	(0.476)	(0.463)	(0.545)
Log(hours) ²	-1.72^{**}	*- 1.68**	*- 1.86**	*- 2.02**	*- 1.32**	*- 1.35**	*- 1.18**	*- 0.98**	*- 1.22**	*- 2.55**	*- 1.48**	*- 1.85**	*- 1.62**	*- 1.65**:	*- 1.03**
	(0.204)	(0.193)	(0.198)	(0.210)	(0.226)	(0.187)	(0.215)	(0.211)	(0.233)	(0.237)	(0.215)	(0.230)	(0.251)	(0.245)	(0.299)
EEA	-0.00	-0.00	0.00	-0.00	0.00	-0.00	0.00	0.00	-0.00	0.00	-0.00	0.00	-0.00	-0.00	0.00
	(0.001)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)
Non-EEA	0.00	-0.00	0.00	-0.00	-0.00	-0.00^{**}	-0.00	-0.00	0.00	-0.00	0.00	0.00	-0.00	-0.00	0.00
	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)
Age	0.06	-0.56	-0.58	-0.04	0.21	0.40	-0.47	0.12	-0.46	-0.77	-0.80	0.85	-0.47	-0.74	0.17
	(0.517)	(0.462)	(0.492)	(0.543)	(0.579)	(0.541)	(0.561)	(0.569)	(0.583)	(0.562)	(0.546)	(0.549)	(0.560)	(0.561)	(0.671)
Age^{2}	-0.08	0.18	0.08	-0.04	-0.17	-0.19	0.17	0.03	0.29	0.22	0.61^{**}	-0.26	0.35	0.57*	-0.16
	(0.250)	(0.225)	(0.238)	(0.266)	(0.289)	(0.266)	(0.275)	(0.281)	(0.288)	(0.283)	(0.277)	(0.279)	(0.290)	(0.291)	(0.346)
Experience	-0.04	0.18	0.18	0.03	-0.05	-0.04	0.17	-0.09	0.10	0.30	0.10	-0.29	0.11	0.02	0.11
	(0.187)	(0.164)	(0.177)	(0.196)	(0.217)	(0.199)	(0.205)	(0.199)	(0.212)	(0.193)	(0.185)	(0.185)	(0.187)	(0.187)	(0.226)
Experience ²	0.08	0.01	0.06	0.06	0.10	0.05	-0.03	0.02	- 0.06	-0.03	-0.15*	0.07	-0.13	-0.09	0.00
	(0.080)	(0.071)	(0.075)	(0.083)	(0.096)	(0.086)	(0.088)	(0.089)	(0.092)	(0.00)	(0.088)	(0.087)	(0.092)	(0.092)	(0.109)

Table A6 con	tinued														
Year	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020
Intermediate educ	0.00	-0.00	0.01	0.00	- 0.00	0.01	- 0.00	-0.01	0.00	0.01	- 0.01	$- 0.02^{*}$	-0.02^{**}	-0.01	-0.01
	(0.007)	(0.007)	(0.008)	(0.008)	(0.00)	(0.008)	(0.00)	(0.00)	(0.009)	(0.009)	(600.0)	(0.000)	(0.009)	(0.010)	(0.011)
High educ	0.00	0.01	0.01	0.00	0.00	0.00	-0.01	-0.02	0.00	0.01	-0.02	-0.04^{**}	-0.03*	-0.03	-0.02
	(600.0)	(0.00)	(0.010)	(0.011)	(0.012)	(0.012)	(0.013)	(0.013)	(0.015)	(0.014)	(0.015)	(0.015)	(0.016)	(0.017)	(0.021)
Years educ	-0.02	-0.09	0.13	0.17	0.21	0.59	0.84^{**}	0.33	1.03^{**}	0.66	0.88^{**}	1.13^{***}	0.60	0.66	1.27^{**}
	(0.318)	(0.316)	(0.332)	(0.358)	(0.394)	(0.381)	(0.402)	(0.415)	(0.429)	(0.429)	(0.435)	(0.436)	(0.459)	(0.477)	(0.539)
Years educ ²	0.01	0.05	-0.03	-0.03	-0.06	-0.21	-0.38^{**}	-0.13	-0.46^{**}	-0.25	-0.36^{*}	-0.47^{**}	-0.19	-0.25	-0.55^{**}
	(0.141)	(0.142)	(0.150)	(0.163)	(0.179)	(0.174)	(0.186)	(0.193)	(0.199)	(0.200)	(0.204)	(0.205)	(0.217)	(0.228)	(0.259)
Training	-0.01	-0.00	0.01	0.00	0.00	-0.00	0.00	0.00	0.00	-0.00	0.00	0.00	0.00	0.00	-0.00
	(0.008)	(0.007)	(0.008)	(0.008)	(0.008)	(0.004)	(0.004)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)
In couple	0.03^{***}	0.02^{**}	0.05***	0.04^{***}	0.01	0.02^{*}	0.05^{***}	0.02	0.04^{***}	0.03^{***}	0.02	0.02	0.04^{***}	0.02^{*}	0.03^{**}
	(0.010)	(0.010)	(0.010)	(0.011)	(0.012)	(0.011)	(0.012)	(0.012)	(0.012)	(0.012)	(0.013)	(0.012)	(0.012)	(0.012)	(0.012)
With dep. children	0.04^{***}	0.03***	0.03***	0.04***	0.04^{***}	0.02**	0.04***	0.03**	0.05***	0.01	0.03^{**}	0.01	0.02*	0.03**	0.02
	(0.010)	(600.0)	(0.010)	(0.011)	(0.011)	(0.010)	(0.011)	(0.012)	(0.012)	(0.012)	(0.012)	(0.012)	(0.012)	(0.013)	(0.010)
In couple with dep. childrei	h — 0.00 n	0.00	-0.00	0.00	- 0.00	0.00	- 0.03***	- 0.00	- 0.03**:	$^{*}-0.01$	-0.00	0.00	-0.01	-0.01	-0.01
	(600.0)	(0.008)	(0.008)	(0.00)	(0.00)	(0.00)	(0.010)	(0.010)	(0.010)	(0.010)	(0.010)	(0.010)	(0.010)	(0.011)	(0.009)
Public sector	-0.00	0.00	-0.00	0.00	-0.00	-0.00	-0.00	-0.01^{**}	-0.00	-0.00	-0.00	0.00	0.00	0.00	-0.00
	(0.002)	(0.002)	(0.002)	(0.002)	(0.003)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)
Constant	- 1.27**:	* - 0.87**:	*- 1.08**	*- 1.79***	* - 0.88**	*- 0.89**	·- 0.81**	-0.39	-0.94^{**}	- 1.91**:	*-0.84**	·- 1.95***	- 1.03**	- 1.50**	$^{*}-1.10^{**}$
	(0.345)	(0.314)	(0.327)	(0.352)	(0.384)	(0.353)	(0.364)	(0.367)	(0.392)	(0.392)	(0.386)	(0.381)	(0.401)	(0.407)	(0.456)

Gender sectoral segregation in the UK

Table A6 con	tinued														
Year	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020
Interaction ef	cect														
Low gender sect. segregation	-0.01	- 0.00	-0.00	- 0.00	- 0.01	0.01	-0.01	0.01	0.01	0.02	0.02	0.01	- 0.00	-0.01	- 0.02**
0	(010.0)	(010.0)	(0.011)	10.01.42	(0.010)	10.015	1010.01	(010.0)	(010.0)	(010.0)		(010.0)	(0.011)	(0.011)	
	(0.012)	(0.010)	(0.011)	(0.014)	(0.013)	(0.010)	(0.012)	(0.012)	(0.010)	(0.010)	(0.012)	(0.010)	(0.011)	(0.011)	(0.012)
Log(hours)	0.27^{***}	0.24^{***}	0.26^{***}	0.32^{***}	0.19^{***}	0.18^{***}	0.14^{***}	0.11^{***}	0.16^{***}	0.39***	0.20^{***}	0.26^{***}	0.21^{***}	0.23^{***}	0.08^{**}
	(0.036)	(0.031)	(0.032)	(0.035)	(0.041)	(0.033)	(0.035)	(0.034)	(0.039)	(0.041)	(0.035)	(0.037)	(0.040)	(0.037)	(0.036)
Log(hours) ²	-0.30^{**}	*- 0.27**	*- 0.30**	**- 0.34**	*- 0.24**	*- 0.25**	*-0.20**	$*-0.16^{**}$	*- 0.21**	*- 0.42**	*- 0.25**	*- 0.30**	$*-0.26^{**}$	*- 0.25**	*-0.13**
	(0.036)	(0.032)	(0.033)	(0.036)	(0.042)	(0.034)	(0.036)	(0.036)	(0.040)	(0.040)	(0.036)	(0.039)	(0.041)	(0.038)	(0.038)
EEA	0.00	0.00	-0.00	0.00	-0.00	0.00	-0.00	-0.00	0.00	-0.00	0.00	-0.00	0.00	0.00	-0.00
	(0.000)	(0.000)	(0.001)	(0.001)	(0.000)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)
Non-EEA	-0.00	0.00	-0.00	0.00	0.00	-0.00	0.00	-0.00	-0.00	0.00	-0.00	-0.00	0.00	0.00	-0.00
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.001)
Age	0.00	-0.04	-0.05	-0.00	0.01	0.02	-0.02	0.01	-0.02	-0.03	-0.03	0.03	-0.01	-0.03	0.01
	(0.037)	(0.036)	(0.038)	(0.034)	(0.027)	(0.022)	(0.027)	(0.026)	(0.024)	(0.023)	(0.022)	(0.019)	(0.018)	(0.021)	(0.029)
Age ²	-0.01	0.03	0.01	-0.00	-0.02	-0.01	0.02	0.00	0.02	0.02	0.05^{**}	-0.02	0.02	0.04^{*}	-0.01
	(0.037)	(0.037)	(0.038)	(0.036)	(0.027)	(0.021)	(0.026)	(0.026)	(0.023)	(0.022)	(0.022)	(0.019)	(0.018)	(0.022)	(0.029)
Experience	-0.01	0.03	0.03	0.00	-0.00	-0.00	0.02	-0.01	0.01	0.02	0.01	-0.02	0.01	0.00	0.01
	(0.026)	(0.026)	(0.027)	(0.024)	(0.020)	(0.015)	(0.020)	(0.019)	(0.018)	(0.016)	(0.016)	(0.015)	(0.014)	(0.016)	(0.021)
Experience ²	0.02	0.00	0.02	0.01	0.01	0.01	-0.00	0.00	-0.01	-0.00	-0.02*	0.01	-0.01	-0.01	0.00
	(0.020)	(0.021)	(0.021)	(0.020)	(0.014)	(0.010)	(0.014)	(0.014)	(0.013)	(0.012)	(0.012)	(0.010)	(0.010)	(0.012)	(0.016)

	וותכת														
Year	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020
Intermediate educ	-0.00	0.00	- 0.00	-0.00	0.00	- 0.00	0.00	0.00	-0.00	-0.00	0.00	0.00	0.00**	0.00	0.00
	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.001)	(0.001)	(0.001)	(0.001)
High educ	0.00	-0.00	-0.00	-0.00	-0.00	-0.00	0.00	0.00	-0.00	-0.00	0.00	0.01^{**}	0.00	0.00	0.00
	(0.000)	(0.001)	(0.000)	(0.000)	(0.000)	(0.000)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.002)	(0.002)	(0.002)	(0.002)
Years educ	-0.00	-0.00	0.00	0.00	0.00	0.01	0.01	-0.00	0.01	-0.00	0.00	-0.00	-0.00	-0.01	-0.01^{*}
	(0.006)	(0.003)	(0.003)	(0.001)	(0.005)	(0.005)	(0.005)	(0.001)	(0.005)	(0.003)	(0.004)	(0.004)	(0.004)	(0.006)	(0.008)
Years educ ²	0.00	0.00	-0.00	-0.00	-0.00	-0.01	-0.01	0.00	-0.00	0.00	-0.00	0.00	0.00	0.01	0.01
	(0.005)	(0.003)	(0.002)	(0.000)	(0.004)	(0.005)	(0.004)	(0.001)	(0.004)	(0.002)	(0.003)	(0.004)	(0.003)	(0.005)	(0.007)
Training	-0.00	-0.00	0.00	0.00	0.00	-0.00	-0.00	0.00	0.00	-0.00	0.00	0.00	-0.00	-0.00	0.00
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
In couple	0.00^{***}	0.00^{**}	0.01^{***}	0.00^{***}	0.00	0.00*	0.01^{***}	0.00	0.01^{***}	0.00^{***}	0.00	0.00	0.00^{**}	0.00*	0.00^{**}
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.002)	(0.001)	(0.002)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.002)
With dep. children	- 0.00*	-0.00	- 0.00	-0.00^{**}	- 0.00	-0.00	- 0.00	-0.00*	- 0.00	- 0.00	- 0.00	-0.00	-0.00	- 0.00	0.00
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.000)	(0.001)	(0.000)	(0.001)	(0.001)	(0.001)
In couple with dep. children	-0.00	0.00	-0.00	0.00	- 0.00	0.00	$- 0.00^{**}$	-0.00	-0.01^{***}	-0.00	-0.00	0.00	-0.00	- 0.00	- 0.00
	(0.001)	(0.001)	(0.001)	(0.001)	(0.002)	(0.001)	(0.002)	(0.001)	(0.002)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)
Public sector	-0.00	-0.00	0.00	-0.00	0.00	0.00	0.00	0.00^{**}	0.00	0.00	0.00	-0.00	-0.00	-0.00	0.00
	(0.000)	(0.000)	(0.000)	(0.000)	(0.001)	(0.000)	(0.000)	(0.001)	(0.001)	(0.001)	(0.001)	(0.000)	(0.001)	(0.001)	(0.000)
Observations	17,046	17,745	16,772	15,574	13,410	13,840	13,204	13,218	13,606	12,994	12,447	12,874	12,085	11,588	10,995
Contribution of levels: p value	main soci << 0.01 ***	o-demogr:	aphic char < 0.05 **,	acteristics, $\frac{1}{p}$ value < 0.	human car	oital attrib	utes and sec	storal indic	ators. Occup	ations (SO	C) are on	itted for]	layout rea	sons. Sig	nificance

Table A6 continued

	Men versus Women in female- dominated sectors	Men versus Women in male- dominated sectors	Workers in female- dominated sectors versus male- dominated sectors if women	Workers in female- dominated sectors versus male- dominated sectors if men
	(1)	(2)	(3)	(4)
Predicted wage	0.1700	0.1541	0.2438	0.1890
Residual wage	0.0366	0.0243	0.0442	0.0231
Counterfactual predicted wage	0.1055	0.0871	0.0867	0.0471
Counterfactual residual wage	0.0821	0.1136	0.1518	0.1791

Table A7 Kolmogorov-Smirnov test for equality of distributions

The Kolmogorov–Smirnov test is a non-parametric test for comparing the equality of two distributions without imposing the normality assumption. The first two columns compare the distribution of men against the distribution of women working in female-dominated sectors (Column 1) or male-dominated sectors (Column 2). The last two columns compare the distribution of workers in female-dominated sectors against those in male-dominated sectors if women (Column 3) or if men (Column 4). All test statistics are significant at 1% level

Appendix B

See Fig. B1.



Fig. B1 Covariate imbalance test, single components. *Note*: The included covariates are balanced because the standardised bias after matching is within $\pm 5\%$ (Rosenbaum and Rubin, 1985). Therefore, the matching method builds a valid comparison group

Appendix C

See Tables C1 and C2.

2 1										
Occupations	1	2	3	4	5	6	7	8	9	Total
Sectors										
A—Agriculture, forestry and fishing	0.09	0.02	0.03	0.05	0.16	50.0	30.01	10.0	50.24	0.69
B—Mining and quarrying	0.08	0.10	0.08	0.04	0.07	70.0	00.00	10.1	0.01	0.48
C—Manufacturing	1.89	1.45	1.55	1.08	2.46	50.0	30.3	12.6	81.13	12.57
D-Electricity, gas and air con supply	0.09	0.14	0.11	0.07	0.13	30.0	00.10	0.0	40.02	0.70
E—Water supply, sewerage and waste	0.11	0.09	0.10	0.08	0.05	50.0	00.03	30.2	20.20	0.89
F—Construction	0.92	0.71	0.45	0.59	1.96	50.02	20.08	30.5	70.49	5.78
G—Distribution	2.17	0.44	1.10	1.34	1.08	30.0	56.30	50.9	51.90	15.38
H—Transport	0.51	0.19	0.42	0.55	0.21	0.3	20.18	31.6	51.06	5.08
I—Accommodation and food services	0.64	0.03	0.13	0.27	0.86	50.1	70.30	00.1	12.90	5.41
J—Information, communication	0.56	1.27	0.65	0.27	0.19	90.0	00.24	40.0	70.28	3.54
K-Financial and insurance services	0.98	0.67	1.25	1.33	0.02	20.0	10.40	0.0	10.06	4.73
L-Real estate services	0.29	0.10	0.33	0.27	0.06	50.0	50.08	80.0	10.05	1.23
M-Professional, scientific and technical activities	\$1.05	1.99	1.37	1.27	0.18	30.1	10.18	80.1	10.42	6.69
N—Admin and support services	0.55	0.40	0.54	0.44	0.25	50.2	50.3	10.14	41.02	3.90
P—Education	0.34	6.18	0.95	0.97	0.19	92.6	80.04	40.0	41.00	12.39
Q-Health and social work	1.14	3.90	2.66	1.78	0.22	25.1	50.13	30.0	90.72	15.79
R-Arts, entertainment and recreation	0.38	0.20	0.51	0.40	0.19	90.2	80.12	20.0	30.36	2.47
S—Other service activities	0.21	0.34	0.23	0.28	0.1	0.6	50.06	50.0	50.18	2.11
T—Households as employers	0.00	0.00	0.00	0.00	0.02	20.0	90.00	0.00	00.03	0.15
Total	11.99	18.21	12.44	411.11	8.4	19.9	08.94	16.9	412.07	7100.00

Table C1 Share of workers by sector and occupation

The percentages are calculated on total employed workers (self-employed are excluded). Labels for occupations: 1—Managers, Directors And Senior Official; 2—Professional Occ.; 3—Associate Professional And Technical Occ.; 4—Administrative And Secretarial Occ.; 5—Skilled Trades Occ.; 6—Caring, Leisure And Other Service Occ.; 7—Sales And Customer Service Occ.; 8—Process, Plant And Machine Operatives; 9—Elementary Occ

full
imposition,
decc
KBO
of
ponents
con
vidual
f indi
ution o
Contrib
le C2
Tab

Table C2 Co	ntribution	of individ	ual compor	nents of KF	30 decomp	osition, fu	ll sample								
Year	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020
Endowment o	sffect														
Log(hours)	-0.11^{**}	*-0.08**	*- 0.15**	*-0.11	*- 0.12**:	*- 0.12**:	*-0.11**	*- 0.06**	*-0.10**	$*-0.16^{***}$	*- 0.12***	$-0.11^{**:}$	*- 0.12**	*- 0.08**	*- 0.07***
	(0.013)	(0.013)	(0.013)	(0.013)	(0.013)	(0.013)	(0.013)	(0.013)	(0.014)	(0.014)	(0.014)	(0.014)	(0.015)	(0.015)	(0.015)
Log(hours) ²	0.11^{***}	0.07^{***}	0.15^{***}	0.11^{***}	0.12^{***}	0.12^{***}	0.10^{***}	0.06^{***}	0.10^{***}	0.16^{***}	0.12^{***}	0.10^{***}	0.12^{***}	0.07^{***}	0.06^{***}
	(0.014)	(0.014)	(0.015)	(0.014)	(0.015)	(0.015)	(0.015)	(0.015)	(0.015)	(0.015)	(0.015)	(0.015)	(0.016)	(0.016)	(0.016)
EEA	-0.00	-0.00	-0.00	-0.00	0.00	0.00^{**}	0.00	0.00*	0.00	0.00^{**}	0.00	0.00^{***}	0.00	0.00	0.00*
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Non-EEA	-0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	-0.00	-0.00**	-0.00^{**}	-0.00	-0.00	-0.00	0.00
	(0.000)	(0.000)	(0.000)	(0.00)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Age	0.01	0.02^{***}	0.03^{**}	-0.00	0.00	0.00	0.01^{**}	-0.00	0.00	0.00	0.00	-0.00	0.00	0.00	0.01^{**}
	(0.007)	(0.008)	(0.010)	(0.00)	(0.003)	(0.002)	(0.004)	(0.003)	(0.002)	(0.003)	(0.002)	(0.002)	(0.002)	(0.002)	(0.005)
Experience	0.01^{**}	0.01^{**}	0.01^{**}	0.03^{***}	0.01^{**}	0.00	0.00	0.01^{***}	0.00^{**}	0.01^{**}	0.01^{**}	0.01^{**}	0.00*	0.01^{**}	0.00
	(0.005)	(0.005)	(0.006)	(0.007)	(0.002)	(0.002)	(0.002)	(0.003)	(0.002)	(0.003)	(0.003)	(0.003)	(0.002)	(0.003)	(0.002)
Years educ	0.01^{***}	0.00	-0.00	-0.01^{**}	0.00	0.00	-0.00	-0.01^{**}	-0.01	-0.01^{***}	*- 0.01	$-0.01^{**:}$	*- 0.02**	*- 0.02**	*- 0.02***
	(0.003)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.005)	(0.004)	(0.004)	(0.004)	(0.004)	(0.005)	(0.005)	(0.005)

Table C2 ot	ontinued														
Year	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020
Training	- 0.00**:	*- 0.00***	- 0.00***	· 0.00***	*-0.00**	*- 0.00***	*- 0.00***	*- 0.00***	*-0.00**	$-0.00^{**:}$	*-0.00	-0.00*	$- 0.00^{*}$	-0.00	$- 0.00^{**}$
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
In couple	-0.00	0.00	-0.00	-0.00	0.00	0.00^{**}	0.00	0.00	0.00	0.00	0.00	0.00^{**}	0.00^{**}	0.00	0.00
	(0.000)	(0.00)	(0000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
With dep. children	0.00***	0.00***	0.00***	0.00***	0.00***	0.00***	0.00***	0.00***	0.00***	0.00***	0.00***	0.00**	0.00***	0.00***	0.00
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.000)	(0.001)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
In couple with dep. children	0.00	- 0.00	0.00	- 0.00	0.00	0.00	0.00	0.00*	0.00**	0.00*	0.00*	0.00	0.00	0.00*	0.00
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.001)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Low gender sect. segre gation	r 0.02*** 3-	0.02***	0.02***	0.02***	0.02***	0.01***	0.02***	0.01***	0.01***	0.01***	0.00*	0.00**	0.01**	0.01***	- 0.00
	(0.005)	(0.004)	(0.005)	(0.004)	(0.005)	(0.004)	(0.004)	(0.004)	(0.003)	(0.003)	(0.003)	(0.002)	(0.003)	(0.003)	(0.003)
Public secto	$r = 0.01^{**}$	$*-0.01^{***}$	-0.01^{***}	*- 0.02***	*- 0.02**:	$*-0.01^{**}$	*- 0.02***	*- 0.02***	*- 0.01***	*- 0.01**:	*-0.01	$^{*}- 0.01^{**}$	*- 0.01**:	*- 0.01**:	$^{*}-0.01^{***}$
	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)

continued	
С	
Ð	
9	
Тa	

 $\underline{\textcircled{O}}$ Springer

	manin														
Year	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020
Coefficient e	zffect														
Log(hours)	1.65^{***}	0.96^{***}	2.02***	2.41***	0.62^{***}	1.51^{***}	1.31^{***}	0.96***	0.49^{**}	2.36***	1.65^{***}	1.56^{***}	1.79^{***}	2.14***	1.22^{***}
	(0.240)	(0.229)	(0.233)	(0.243)	(0.226)	(0.219)	(0.239)	(0.242)	(0.239)	(0.263)	(0.260)	(0.270)	(0.302)	(0.296)	(0.341)
Log(hours) ²	2 - 0.96**	*- 0.63**	*- 1.19**	**- 1.34**	*- 0.50**:	*- 0.94***	*- 0.83***	*- 0.66**:	*- 0.39**:	*- 1.31***	·- 0.98***	*- 0.93***	*- 1.06**3	*- 1.21**	*- 0.82***
	(0.122)	(0.120)	(0.123)	(0.129)	(0.121)	(0.117)	(0.127)	(0.128)	(0.127)	(0.138)	(0.137)	(0.141)	(0.157)	(0.157)	(0.183)
EEA	-0.00	0.00	0.00	-0.00	-0.00	0.00	0.00	0.00	-0.00	-0.00	-0.00	0.00	-0.00	0.00	-0.00
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)
Non-EEA	0.00	0.00	0.00	0.00	-0.00	-0.00^{**}	-0.00	-0.00	-0.00	-0.00	0.00	-0.00	-0.00	-0.00	0.00
	(0.001)	(0.001)	(0.001)	(0.001)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)
Age	-0.04	0.31	-0.03	0.83^{***}	0.33	0.49	-0.08	0.89^{***}	0.23	-0.06	0.03	0.84^{**}	-0.33	0.27	0.06
	(0.303)	(0.288)	(0.304)	(0.313)	(0.342)	(0.327)	(0.330)	(0.336)	(0.336)	(0.343)	(0.326)	(0.339)	(0.338)	(0.350)	(0.420)
Experience	0.29^{**}	0.02	0.13	-0.17	0.00	0.19	0.18	-0.09	0.10	0.13	-0.00	-0.25^{**}	0.15	-0.10	0.19
	(0.117)	(0.107)	(0.115)	(0.118)	(0.133)	(0.125)	(0.123)	(0.124)	(0.123)	(0.124)	(0.112)	(0.119)	(0.116)	(0.118)	(0.146)
Years educ	0.32	0.36^{*}	0.23	0.39*	0.31	0.42*	0.45*	0.26	0.80^{***}	0.80^{***}	0.80^{***}	0.81^{***}	0.30	0.82^{***}	0.66^{*}
	(0.198)	(0.202)	(0.213)	(0.222)	(0.242)	(0.236)	(0.250)	(0.254)	(0.258)	(0.268)	(0.271)	(0.274)	(0.284)	(0.300)	(0.344)

Table C2 ot	ontinued														
Year	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020
Training	0.01^{**}	0.01	0.01^{*}	0.02^{***}	0.01	-0.00	0.00*	0.00	0.01^{***}	0.00	0.01^{***}	0.01^{***}	0.00	0.01^{**}	0.00
	(0.005)	(0.005)	(0.005)	(0.006)	(0.005)	(0.003)	(0.003)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)
In couple	0.03^{***}	0.02^{***}	0.04^{***}	0.03^{***}	0.02^{**}	0.01	0.04^{***}	0.02^{***}	0.04^{***}	0.04^{***}	0.04^{***}	0.03^{***}	0.03^{***}	0.02^{***}	0.04^{***}
	(0.007)	(0.007)	(0.007)	(0.008)	(0.008)	(0.008)	(0.008)	(0.008)	(0.008)	(0.008)	(0.008)	(0.008)	(0.008)	(0.008)	(0.008)
With dep. children	0.02^{***}	0.03***	0.04***	0.03***	0.04***	0.03***	0.04***	0.03***	0.04***	0.03***	0.03***	0.02***	0.03***	0.04***	0.01*
	(0.007)	(0.007)	(0.007)	(0.008)	(0.008)	(0.007)	(0.008)	(0.008)	(0.008)	(0.008)	(0.008)	(0.008)	(0.008)	(0.008)	(0.007)
In couple with dep. children	0.01	0.01	0.00	0.00	-0.00	0.00	- 0.02***	*0.01	-0.01^{**}	-0.01*	- 0.01	- 0.00	-0.00	- 0.01	- 0.00
	(0.006)	(0.006)	(0.006)	(0.006)	(0.006)	(0.006)	(0.006)	(0.006)	(0.006)	(0.006)	(0.006)	(0.006)	(0.006)	(0.007)	(0.006)
Low gender sect. segre gation	- 0.02 3-	0.00	0.02	0.01	0.04	- 0.03	0.04	- 0.02	- 0.04	- 0.04	-0.05	- 0.04	0.00	0.03	0.07**
	(0.028)	(0.025)	(0.027)	(0.029)	(0.034)	(0.034)	(0.028)	(0.029)	(0.027)	(0.029)	(0.035)	(0.030)	(0.030)	(0.029)	(0.031)
Public sectc	r = 0.01 **	$^{*-} 0.01^{**}$	-0.02^{**}	*-0.01	*-0.01 **:	$^{*}-0.01$	-0.02^{***}	*- 0.02**	*- 0.02**:	$^{*}- 0.01^{*}$	-0.01^{**}	*-0.00	-0.00	-0.01^{**}	-0.01^{**}
	(0.004)	(0.004)	(0.005)	(0.005)	(0.005)	(0.005)	(0.005)	(0.005)	(0.005)	(0.005)	(0.005)	(0.005)	(0.005)	(0.005)	(0.006)
Constant	-0.89^{**}	*- 0.65**	*- 0.98**	*- 1.61**	*- 0.50**	-0.80^{**}	*- 0.83***	*- 0.77**	*- 0.57**	-1.46^{**}	*- 0.93**	*- 1.30**	*- 0.83**:	*- 1.64**:	·- 1.08***
	(0.214)	(0.203)	(0.211)	(0.220)	(0.232)	(0.229)	(0.227)	(0.230)	(0.232)	(0.240)	(0.253)	(0.249)	(0.256)	(0.262)	(0.296)

 $\underline{\textcircled{O}}$ Springer

continue	
Q	
Table	

Table C2 ot	ntinued														
Year	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020
Interaction	effect														
Log(hours)	0.19^{***}	0.11^{***}	0.22^{***}	0.25^{***}	0.07^{***}	0.17^{***}	0.14^{***}	0.10^{***}	0.05^{**}	0.23^{***}	0.16^{***}	0.15^{***}	0.17^{***}	0.19^{***}	0.10^{***}
	(0.028)	(0.025)	(0.025)	(0.026)	(0.026)	(0.025)	(0.026)	(0.026)	(0.025)	(0.026)	(0.026)	(0.027)	(0.029)	(0.027)	(0.027)
Log(hours) ²	$^2 - 0.22^{**}$	*-0.14**	*- 0.25**	**- 0.27**	*-0.11**	*- 0.21**	*- 0.17**	*-0.13**	*- 0.08**	*- 0.25**:	*- 0.19**:	*- 0.18***	*- 0.20**:	*- 0.21**:	- 0.12***
	(0.028)	(0.026)	(0.026)	(0.026)	(0.027)	(0.026)	(0.026)	(0.026)	(0.025)	(0.027)	(0.026)	(0.027)	(0.029)	(0.027)	(0.028)
EEA	-0.00	0.00	-0.00	0.00	0.00	-0.00	-0.00	-0.00	0.00	0.00	0.00	-0.00	0.00	-0.00	0.00
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Non-EEA	0.00	0.00	0.00	0.00	-0.00	-0.00^{**}	-0.00	-0.00	-0.00	-0.00	0.00	-0.00	-0.00	-0.00	-0.00
	(0.00)	(0.00)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.00)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Age	-0.00	0.01	-0.00	0.04^{***}	0.00	0.00	-0.00	0.01^{**}	0.00	-0.00	0.00	0.01	-0.00	0.00	0.00
	(0.010)	(0.011)	(0.014)	(0.014)	(0.005)	(0.003)	(0.004)	(0.005)	(0.003)	(0.003)	(0.003)	(0.004)	(0.002)	(0.003)	(0.005)
Experience	0.02^{**}	0.00	0.01	-0.01	0.00	0.00	0.00	-0.00	0.00	0.00	-0.00	-0.00	0.00	-0.00	0.00
	(0.007)	(0.008)	(0.009)	(0.009)	(0.003)	(0.002)	(0.003)	(0.003)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.004)
Years educ	0.00	0.00	-0.00	-0.00	0.00	0.00	-0.00	-0.00	-0.00	-0.01^{**}	-0.00	-0.01^{**}	-0.00	-0.01^{**}	-0.01*
	(0.002)	(0.001)	(0.001)	(0.002)	(0.001)	(0.001)	(0.001)	(0.002)	(0.002)	(0.003)	(0.002)	(0.003)	(0.003)	(0.004)	(0.005)

Table C2 o	ontinued														
Year	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020
Training	-0.00*	-0.00	-0.00*	-0.00^{**}	- 0.00	0.00	-0.00*	- 0.00	-0.00^{**}	- 0.00	-0.00^{**}	-0.00*	- 0.00	-0.00	- 0.00
	(0.000)	(0.00)	(0.000)	(0.000)	(0.000)	(0.000)	(0.00)	(0.00)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
In couple	0.00^{***}	0.00^{**}	0.00^{***}	0.00^{***}	0.00*	0.00	0.00^{***}	0.00^{**}	0.00^{***}	0.00^{***}	0.00^{***}	0.00^{***}	0.00^{***}	0.00^{**}	0.00^{***}
	(0.000)	(0.00)	(0.001)	(0.000)	(0.000)	(0.000)	(0.001)	(0.00)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)
With dep. children	- 0.00**	*- 0.00**	*= 0.00**	*- 0.00**:	*- 0.00**	*- 0.00**	*- 0.00**	*- 0.00**	*- 0.00**	*- 0.00**	*- 0.00***	·- 0.00**	*- 0.00**	*- 0.00**	*-0.00
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.000)
In couple with dep. children	0.00	- 0.00	0.00	- 0.00	- 0.00	0.00	- 0.00**	0.00	- 0.00*	- 0.00*	- 0.00	- 0.00	- 0.00	- 0.00	- 0.00
	(0.000)	(0.00)	(0.000)	(0.000)	(0.000)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)
Low gender sect. segra gation	r — 0.00	- 0.00	- 0.00	- 0.00	-0.01	0.00	- 0.00	0.00	0.00	0.00	0.00	0.00	- 0.00	- 0.00	0.00
	(0.005)	(0.005)	(0.005)	(0.004)	(0.004)	(0.004)	(0.004)	(0.003)	(0.003)	(0.003)	(0.003)	(0.002)	(0.002)	(0.002)	(0.002)
Public secto	or 0.01***	0.01^{**}	0.01^{***}	0.01^{***}	0.01^{***}	0.00	0.01^{***}	0.01^{***}	0.01^{***}	0.01^{*}	0.01^{***}	0.00	0.00	0.01^{**}	0.01^{**}
	(0.002)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)
Observation	1s32,066	33,053	31,683	29,746	26,656	27,577	26,540	26,531	27,039	25,873	24,915	25,819	24,236	23,151	21,378
Contributio < 0.1 *	n of main s	ocio-demo	graphic ch	aracteristics	i, human ca	ıpital attrib	utes and se	ctoral indi	cators. Sign	ificance le	vels: <i>p</i> value	s < 0.01 *	***, <i>p</i> valu	e < 0.05 **	[*] , <i>p</i> value

Gender sectoral segregation in the UK

Appendix D

See Fig. D1.



Fig. D1 KBO decomposition, full sample. *Note*: The figure shows the evolution of the components of the KBO decomposition and their sum over time for full sample. Women are contrasted to men. The dashed line represents the *coefficient effect*, the long-dashed line the *endowment effect* and the dotted line the part of the "unexplained" component of the threefold decomposition (or *interaction effect*). The corresponding shadowed areas display the 95% confidence intervals. The solid line is the sum of the three effects and reveals their overall contribution. The wage difference between men and women is, on average, 0.2 logarithmic points over time. Most of the gender pay gap (around three-fourths) can be explained by differences in the estimated coefficients between genders. The *coefficient effect* on average quantifies indeed an increase of 0.16 logarithmic points in women's wages when the male coefficients are applied to female characteristics. In addition, this component displays a downward trend after 2008. The *endowment effect* would quantify an expected average increase in women's wage by around 0.05 points if they had male predictors levels. Therefore, differences in observed characteristics account for one-fourth of the gap. ESTIMATION NOTE: Both models for women and men are estimated using the Mincerian regression equation (with OLS). The shaded area is the 95% confidence intervals

References

- Aaron HJ, Lougy MC (1987) The comparable worth controversy. The Brookings Institute, Washington, DC Albanesi S, Kim J (2021) Effects of the covid-19 recession on the us labor market: occupation, family, and gender. J Econ Perspect 35(3):3–24
- Altonji JG, Blank RM (1999) Race and gender in the labor market. Handbook Labor Econ 3:3143-3259
- Bertoli S, Dequiedt V, Zenou Y (2016) Can selective immigration policies reduce migrants' quality? J Dev Econ 119:100–109
- Bertrand M (2011) New perspectives on gender. In: Handbook of Labor Economics, vol 4, Elsevier, pp 1543–1590
- Bertrand M (2020) Gender in the twenty-first century. In: AEA Papers and Proceedings, volume 110, American Economic Association 2014 Broadway, Suite 305, Nashville, TN 37203 pp 1–24
- Bielby WT, Baron JN (1986) Sex segregation within occupations. Am Econ Rev 76(2):43-47
- Blackburn RM, Jarman J, Siltanen J (1993) The analysis of occupational gender segregation over time and place: considerations of measurement and some new evidence. Work Employ Soc 7(3):335–362
- Blau FD, Kahn LM (2017) The gender wage gap: extent, trends, and explanations. J Econ Lit 55(3):789–865 Blinder AS (1973) Wage discrimination: reduced form and structural estimates. J Hum Resour, pp 436–455 Booth AL (2009) Gender and competition. Labour Econ 16(6):599–606
- Borjas GJ (1980) The relationship between wages and weekly hours of work: the role of division bias. J Hum Resour 15(3):409-423
- Borjas GJ (1987) Self-selection and the earnings of immigrants. Am Econ Rev, pp 531-553
- Borjas GJ, Kauppinen I, Poutvaara P (2019) Self-selection of emigrants: theory and evidence on stochastic dominance in observable and unobservable characteristics. Econ J 129(617):143–171
- Campos-Soria JA, Ropero-García MA (2016) Gender segregation and earnings differences in the Spanish labour market. Appl Econ 48(43):4143–4155
- Card D, Cardoso AR, Kline P (2016) Bargaining, sorting, and the gender wage gap: quantifying the impact of firms on the relative pay of women. Quart J Econ 131(2):633–686
- Card D, Colella F, Lalive R (2021) Gender preferences in job vacancies and workplace gender diversity. Working paper 29350, National Bureau of Economic Research
- Carvalho I, Costa C, Lykke N, Torres A (2019) Beyond the glass ceiling: gendering tourism management. Ann Tour Res 75:79–91
- Chiswick B (1999) Are immigrants favorably self-selected? Am Econ Rev 89(2):181-185
- Chiswick BR (1978) The effect of Americanization on the earnings of foreign-born men. J Polit Econ 86(5):897–921
- Chiswick BR (1986) Is the new immigration less skilled than the old? J Law Econ 4(2):168–192
- Cortés P, Pan J (2018) Occupation and gender. The Oxford Handbook of Women and the Economy, pp 425–452
- Cortés P, Pan J (2020) Children and the remaining gender gaps in the labor market. Working paper 27980, National Bureau of Economic Research
- Cortese CF, Falk RF, Cohen JK (1976) Further considerations on the methodological analysis of segregation indices. American Sociological Review, pp 630–637
- Cotton J (1988) On the decomposition of wage differentials. The Review of Economics and Statistics, pp 236–243
- Doepke M, Tertilt M (2016) Families in macroeconomics. In: Handbook of Macroeconomics, vol 2, chapter 3, Elsevier, pp 1789–1891.
- Duncan OD, Duncan B (1955) A methodological analysis of segregation indexes. Am Sociol Rev 20(2):210–217
- Dustmann C, Frattini T (2014) The fiscal effects of immigration to the UK. Econ J 124(580):F593-F643
- Dustmann C, Glitz A, Vogel T (2010) Employment, wages, and the economic cycle: differences between immigrants and natives. Eur Econ Rev 54(1):1–17
- Ehrenberg RG, Smith RS (1987) Comparable worth in the public sector. In: Public Sector Payrolls, University of Chicago Press. pp 243–290
- Ellieroth K, et al (2019) Spousal insurance, precautionary labor supply, and the business cycle-a quantitative analysis. In: Society for Economic Dynamics 2019 Meeting Papers, vol 1134
- Ferber MA (1986) What is the worth of comparable worth? J Econ Educat 17(4):267-282
- Folke O, Rickne J (2022) Sexual harassment and gender inequality in the labor market. Quart J Econ 137(4):2163–2212

Gerhart B (1991) Equity and gender: the comparable worth debate. Adm Sci Q 36(2):312–314

- Gleason SE (1985) Comparable worth: some questions still unanswered. Mon Labor Rev 108:17
- Gneezy U, Niederle M, Rustichini A (2003) Performance in competitive environments: gender differences. Quart J Econ 118(3):1049–1074
- Gneezy U, Rustichini A (2004) Gender and competition at a young age. Am Econ Rev 94(2):377-381
- Goldin C (2014) A grand gender convergence: its last chapter. Am Econ Rev 104(4):1091-1119
- Gould ED, Moav O (2016) Does high inequality attract high skilled immigrants? Econ J 126(593):1055–1091
- Government Equalities Office (2019) Gender equality monitor: Tracking progress on gender equality. Report, HM Government
- Hoynes H, Miller DL, Schaller J (2012) Who suffers during recessions? J Econ Perspect 26(3):27-48
- Irvine S (2022) Women and the UK economy. https://researchbriefings.files.parliament.uk/documents/ SN06838/SN06838.pdf [Accessed: 17/10/2022]
- James DR, Taeuber KE (1985) Measures of segregation. Sociol Methodol 15:1-32
- Jann B (2018) Decomposition methods in the social sciences. University of Bern, Institut of Sociology. [Accessed: 19/11/2023 at https://boris.unibe.ch/117107/4/8-did.pdf
- Jann B et al (2008) A stata implementation of the blinder-oaxaca decomposition. Stand Genomic Sci 8(4):453–479
- Jewell SL, Razzu G, Singleton C (2020) Who works for whom and the UK gender pay gap. Br J Ind Relat 58(1):50–81
- Johnson GE, Solon G (1984) Pay differences between women's and men's jobs: The empirical foundations of comparable worth legislation. Working paper number 1472, National Bureau of Economic Research
- Johnston I (2021) UK risks 'turning clock back' on gender equality in pandemic. [Online; posted 9-February-2021]
- Kamerāde D, Richardson H (2018) Gender segregation, underemployment and subjective well-being in the UK labour market. Hum Relat 71(2):285–309
- Killingsworth MR (1987) Heterogeneous preferences, compensating wage differentials, and comparable worth. Quart J Econ 102(4):727–742
- Kitagawa EM (1955) Components of a difference between two rates. J Am Stat Assoc 50(272):1168-1194
- Kreimer M (2004) Labour market segregation and the gender-based division of labour. Eur J Women's Stud 11(2):223–246
- Maahs DE, Morrow PC, McElroy JC (1985) Earnings gap in the 1980's: Its causes, consequences and prospects for elimination. Woman CPA 47(1):5
- Madden JF (1987) Comparable worth. J Policy Anal Manage 7(1):147-150
- Meara K, Pastore F, Webster A (2020) The gender pay gap in the USA: a matching study. J Popul Econ 33(1):271–305
- Moir H, Smith JS (1979) Industrial segregation in the Australian labour market. J Ind Relat 21(3):281-291
- Morchio I, Moser C (2021) The gender pay gap: Micro sources and macro consequences. CEPR Discussion Paper No. DP16383
- Mueller G, Plug E (2006) Estimating the effect of personality on male and female earnings. ILR Rev 60(1):3–22
- Mumford K, Smith PN (2008) What determines the part-time and gender earnings gaps in Britain: evidence from the workplace. Oxf Econ Pap 61(suppl-1):i56–i75
- Oaxaca R (1973) Male-female wage differentials in urban labor markets. International Economic Review, pp 693–709
- Office for National Statistics (2022) JOBS02: Workforce jobs by industry. https://www.ons. gov.uk/employmentandlabourmarket/peopleinwork/employmentandemployeetypes/datasets/ workforcejobsbyindustryjobs02 [Accessed: 17/10/2022]
- Olivetti C, Petrongolo B (2014) Gender gaps across countries and skills: demand, supply and the industry structure. Rev Econ Dyn 17(4):842–859
- Olivetti C, Petrongolo B (2016) The evolution of gender gaps in industrialized countries. Ann Rev Econ 8:405–434
- Olsen W, Gash V, Sook K, Zhang M (2018) The gender pay gap in the UK: Evidence from the UKHLS. Research Report Number DFE- RR804, London: Department for Education, Government Equalities Office. Creative Commons
- Open Society Foundations (2020) Q&A: How migrant women in the UK are joining together in the era of COVID-19

Parey M, Ruhose J, Waldinger F, Netz N (2017) The selection of high-skilled emigrants. Rev Econ Stat 99(5):776–792

Périvier H (2014) Men and women during the economic crisis. Revue de l'OFCE 2:41-84

- Pető R, Reizer B (2021) Gender differences in the skill content of jobs. J Popul Econ 34(3):825-864
- Petrongolo B (2004) Gender segregation in employment contracts. J Eur Econ Assoc 2(2-3):331-345
- Piłatowska M, Witkowska D (2022) Gender segregation at work over business cycle—evidence from selected EU countries. Sustainability 14(16):10202
- Razzu G, Singleton C (2018) Segregation and gender gaps in the United Kingdom's great recession and recovery. Fem Econ 24(4):31–55
- Razzu G, Singleton C, Mitchell M (2020) On why the gender employment gap in Britain has stalled since the early 1990s. Ind Relat J 51(6):476–501
- Reardon SF, Firebaugh G (2002) Measures of multigroup segregation. Sociol Methodol 32(1):33-67
- Reuben E, Wiswall M, Zafar B (2017) Preferences and biases in educational choices and labour market expectations: Shrinking the black box of gender. Econ J 127(604):2153–2186
- Roos PA (1981) Sex stratification in the workplace: male-female differences in economic returns to occupation. Soc Sci Res 10(3):195–224
- Rosenbaum PR, Rubin DB (1985) Constructing a control group using multivariate matched sampling methods that incorporate the propensity score. Am Stat 39(1):33–38
- Rubery J (2010) Women and Recession (Routledge Revivals). Routledge, London
- Rubery J, Rafferty A (2013) Women and recession revisited. Work Employ Soc 27(3):414-432
- Saccardo S, Pietrasz A, Gneezy U (2018) On the size of the gender difference in competitiveness. Manage Sci 64(4):1541–1554
- Scarborough WJ, Sobering K, Kwon R, Mumtaz M (2021) The costs of occupational gender segregation in high-tech growth and productivity across us local labor markets. Socio-Economic Review
- Treiman DJ, Hartmann HI et al (1981) Women, work, and wages: Equal pay for jobs of equal value. National Academy Press, Washington, DC
- United States Commission on Civil Rights (1984) Comparable Worth: Issue for the 80's: a Consultation of the U.S. Commission on Civil Rights. Number v. 1
- Watts M (1992) How should occupational sex segregation be measured? Work Employ Soc 6(3):475-487
- Watts M (1995) Divergent trends in gender segregation by occupation in the united states: 1970–92. J Post Keynes Econ 17(3):357–378
- Watts M (1998) Occupational gender segregation: index measurement and econometric modeling. Demography 35(4):489–496

Zoloth BS (1976) Alternative measures of school segregation. Land Econ 52(3):278-298

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Authors and Affiliations

Riccardo Leoncini^{1,2} · Mariele Macaluso¹ · Annalivia Polselli³

Mariele Macaluso mariele.macaluso2@unibo.it; mariele.macaluso@unito.it

Riccardo Leoncini riccardo.leoncini@unibo.it

Annalivia Polselli annalivia.polselli@essex.ac.uk

- ¹ Department of Legal Studies, Alma Mater Studiorum, University of Bologna, Bologna, Italy
- ² IRCrES, National Research Council, Milan, Italy
- ³ Department of Economics, University of Essex, Colchester, United Kingdom